# UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-0

[X] QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES AND EXCHANGE ACT OF 1934

For the quarterly period ended March 29, 1997

OR

[ ] TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d)
OF THE SECURITIES AND EXCHANGE ACT OF 1934

For the transition period from ......to.....

Commission file number 1-14092

THE BOSTON BEER COMPANY, INC. (Exact name of registrant as specified in its charter)

MASSACHUSETTS 04-3284048 (State or other jurisdiction of incorporation or organization) (I.R.S. Employer Identification No.)

75 Arlington Street, Boston, Massachusetts 02116 (Address of principal executive offices) (Zip Code)

(617) 368-5000

Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days.

Yes X No

As of May 2, 1997, there were 16,224,755 shares outstanding of the Registrant's Class A Common Stock (\$.01 par value) and 4,107,355 shares outstanding of the Company's Class B Common Stock (\$.01 par value).

THE BOSTON BEER COMPANY, INC.

INDEX

# PART I. FINANCIAL INFORMATION

Item 1. Financial Statements

Consolidated Balance Sheets March 29, 1997 and December 28, 1996

Consolidated Statements of Income for the Three months ended March 29, 1997 and March 30, 1996

Consolidated Statements of Cash Flows for the Three months ended March 29, 1997 and March 30, 1996

Notes to Consolidated Financial Statements

# Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

# PART II. OTHER INFORMATION

Item 1. Legal Proceedings

Item 2. Changes in Securities

Item 3. Defaults Upon Senior Securities

Item 4. Submission of Matters to a Vote of Security Holders

Item 5. Other Information

Item 6. Exhibits and Reports on Form 8-K

Exhibit Index

Exhibit 10.23

Exhibit 11

#### SIGNATURES

THE BOSTON BEER COMPANY, INC.
CONSOLIDATED BALANCE SHEETS
(Unaudited)
(in thousands, except share data)

	]	March 29, 1997	Dec	cember 28, 1996	
ASSETS					
Current Assets:					
Cash and cash equivalents	\$	32	\$	5,060	
Short term investments		35,536		35,926	
Accounts receivable		21,089		18,109	
Allowance for doubtful accounts		(1,953)		(1,930)	
Inventories		14,254		13,002	
Prepaid expenses		772		674	
Deferred income taxes		2,968		2,968	
Other current assets	_	2 <b>,</b> 699		3,882	
Total current assets		75 <b>,</b> 397		77,691	
Restricted investments Equipment and leasehold		627		611	
improvements, at cost		29,069		21,043	
Accumulated depreciation		(7,378)		(6,412)	
Deferred income taxes		151		151	
Other assets		3,571		3,469	
Total assets	s ·	101,437	: \$	96,553	
10041 400000		=======		======	
LIABILITIES AND STOCKHOLDERS' EQUITY					
Current Liabilities:					
Accounts payable	\$	12,373	\$	17,783	
Notes payable - Bank		9,871		-	
7			11 70	2	10 064
Accrued expenses			11,739		12,064
Current maturities oflong-term del	ot		7 ! 	5 	75 
Total current liabilities			34,058	3	29,922
Long-term debt, less current maturi	ci∈	es	1,800	)	1,800

Stockholders' Equity:		
Class A Common Stock, \$.01 par value;		
20,300,000 shares authorized;		
15,969,840, and 15,972,058		
issued and outstanding as of		
March 28, 1997 and December 28,		
1996, respectively	160	160
Class B Common Stock, \$.01 par value;		
4,200,000 shares authorized;		
4,107,355 issued and outstanding as		
of March 28, 1997 and December 28, 1996	41	41
Additional paid-in-capital	55,410	55 <b>,</b> 391
Unearned compensation	(324)	(363)
Unrealized loss on investments in		
marketable securities	(1,223)	(442)
Unrealized gains (loss) on		
forward exchange contract	(78)	31
Retained earnings	11,593	10,013
Total stockholders' equity	65 <b>,</b> 579	64,831
Total liabilities and		
stockholders' equity	\$ 101,437	\$ 96,553
	=======	========

The accompanying notes are an integral part of the financial statements.

For the Three Months Ended

THE BOSTON BEER COMPANY, INC.
CONSOLIDATED STATEMENTS OF INCOME
(Unaudited)
(in thousands, except per share data)

	March 29, 1997	March 30, 1996
Sales Less excise taxes	\$ 46,799 4,944	\$ 48,276 5,147
Net sales Cost of sales	41,855 21,907	43,129 21,865
Gross profit	19,948	21,264
Operating expenses: Advertising, promotional and selling expenses General and administrative expenses	14,558 2,930	14,029 2,983
Total operating expenses	17,488	17,012
Operating income	2,460	4,252
Other income (expense): Interest income Interest expense Other income, net	451 (108) 7	498 (57) (7)
Total other income	350	434
Income before income taxes Provision for income taxes	2,810 1,230	4,686 2,046
Net income	\$ 1,580 ======	\$ 2,640 ======

Net income per common and

common	equivalent share	\$	0.08	Ş	0.13
		===		=	
Weighted	average number of				
common	and common equivalent shares		20,313		20,366
		===		=	

The accompanying notes are an integral part of the financial statements.

# THE BOSTON BEER COMPANY, INC. CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited) (in thousands)

	For the Quarter Ended		
		March 30, 1996	
CASH FLOWS FOR OPERATING ACTIVITIES: Net income Adjustments to reconcile net income to net cash provided by operating activities:	\$ 1,580	\$ 2,640	
Depreciation and amortization Bad debt expense Stock option compensation expense Changes in assets & liabilities:	966 39 52	539 19 78	
Accounts receivable Inventory Prepaid expenses Other current assets Other assets Accounts payable	(2,996) (1,252) (98) (489) (102) (5,410)	(8,402) (3,428) 58 1,393 1,262 5,394	
Accrued expenses  Total adjustments	(325)  (9,615)	(939)  (4,026)	
Net cash used by operating activities	(8,035)	(1,386)	
CASH FLOWS FOR INVESTING ACTIVITIES: Acquisitions and purchases of fixed assets (Purchases) maturities of government securities Purchases of restricted investments Maturities of restricted investments	(6,463) (390) (625) 609	(2,845) (427) (614) 597	
Net cash used in investing activities	(6,869)	(3,289)	
CASH FLOWS FROM FINANCING ACTIVITES: Proceeds from sale under stock purchase plan Net borrowings under line of credit  Net cash provided by financing activities	9,871 	240 240	
Net increase (decrease) in cash and cash equivalents	(5,028)	(4,435)	
Cash and cash equivalents at beginning of period	5,060	1,877	
Cash and cash equivalents at end of period	\$ 32 ======	\$ (2,558) ======	
Supplemental disclosure of cash flow information: Interest paid Taxes paid	\$ 112 \$ 385	\$ 113 \$ 424	

The accompanying notes are an integral part of the financial statements.

THE BOSTON BEER COMPANY, INC. NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

The accompanying consolidated financial statements have been prepared by the Company, without audit, in accordance with generally accepted accounting principles for interim financial information and pursuant to the rules and regulations of the Securities and Exchange Commission. Accordingly, they do not include all of the information and footnotes required by generally accepted accounting principles for complete financial statements and should be read in conjunction with the audited financial statements included in the Company's Annual Report on Form 10-K for the year ended December 28, 1996.

In the opinion of management, the accompanying unaudited financial statements contain all adjustments, consisting only of those of a normal recurring nature, considered necessary for a fair presentation of the Company's financial position, results of operations and cash flows at the dates and for the periods presented. The operating results for the interim periods presented are not necessarily indicative of the results expected for the full year.

#### Reclassifications

Beginning in 1996, certain expenses which were previously classified as general and administrative expenses were reclassified as advertising, promotional, and selling expenses. All prior period financial information has been reclassified to conform with this year's presentation. Certain other prior period amounts have also been reclassified to conform with the current year's presentation.

### B. SHORT TERM INVESTMENTS:

Short term investments include marketable equity securities having a cost of \$4,286,000 and \$4,286,000 and a market value of \$3,062,500 and \$3,844,000 at March 29, 1997 and December 28, 1996, respectively. This resulted in an unrealized loss of \$1,223,500 and \$442,000 at March 29, 1997 and December 28, 1996, respectively. In addition, the Company has investments in U.S. Government securities having a cost of \$32,473,000 and \$32,082,000 at March 29, 1997 and December 28, 1996, respectively, which approximate fair value.

# C. INVENTORIES:

Inventories, which consist principally of hops, bottles, and packaging, are stated at the lower of cost, determined on a first-in, first-out (FIFO) basis, or market.

Inventories consist of the following (in thousands):

	March 29, 1997	December 28, 1996
Raw materials, principally hops Work in process Finished goods	\$ 13,802 264 188	\$ 12,677 - 325
	\$ 14,254 =======	\$ 13,002 =======

THE BOSTON BEER COMPANY, INC.
NOTES TO CONSOLIDATED FINANCIAL STATEMENTS
(Continued)

# D. FINANCIAL INSTRUMENTS:

During the first quarter of 1997, the Company entered into several forward exchange contracts to reduce exposure to currency movements affecting existing foreign currency denominated assets, liabilities, and firm commitments. The contract durations match the durations of the currency position. The future value of the contracts and the

related currency positions are subject to offsetting market risk resulting from foreign currency exchange rate volatility. The combined carrying amounts of the contracts and the combined unrealized losses recognized as a component of Stockholders' Equity totaled \$5,300,925 and \$77,679, respectively, at March 29, 1997. There were no realized gains or losses on the contracts in the three months ended March 29, 1997.

#### E. ACQUISITION OF THE SCHOENLING BREWERY:

On March 1, 1997 the Company acquired all of the equipment and other brewery-related personal property from The Schoenling Brewing Company located in Cincinnati, Ohio, and leased the real estate on which the brewery is situated. The Cincinnati facility brews alcoholic beverages for The Boston Beer Company, Inc. as well as contract-brews for several independent companies. The results of operations of The Schoenling Brewing Company (d.b.a. The Samuel Adams Brewery Company) are included in the accompanying financial statements since the date of acquisition. The acquisition is being accounted for under the purchase method of accounting. The total purchase price, subject to certain purchase price adjustments, is approximately \$4,433,000. When finalized, the purchase price will be allocated to the assets acquired based on their fair market values. Any excess in costs of the net assets over the purchase price will be recorded as goodwill and will be amortized on a straight-line basis over the average estimated useful life of the assets acquired.

In addition, the Company has acquired an option to purchase the real estate on which the Schoenling brewery is located. As of March 29, 1997, the Company has paid approximately \$1,700,000 for this option. This amount is included in fixed assets at March 29, 1997.

#### F. DEBT

On March 21, 1997, the Company entered into a credit agreement to increase their existing \$14,000,000 line of credit to \$15,000,000 ("the \$15,000,000 line") and to establish an additional \$30,000,000 line of credit ("the \$30,000,000 line"). On March 31, 1999, the \$15,000,000 line expires and the balance outstanding under the \$30,000,000 line converts to a term note. Principal payments on the term note are payable in twenty quarterly installments, with the final payment due at maturity, December 31, 2003. Through March 31, 1999, interest is payable quarterly on both the \$15,000,000 and \$30,000,000 lines at either the Prime Rate plus .50% or the applicable Adjusted Libor Rate plus .25%. After March 31, 1999, interest on the term note is payable quarterly at either the Prime Rate plus .50% or the applicable Adjusted Libor Rate plus .50% or the applicable Adjusted Libor Rate plus .50% or the

At March 29, 1997, \$9,871,000 and \$0 are outstanding under the \$15,000,000 and \$30,000,000 lines, respectively, at an interest rate of 8.5%. The Company must pay a commitment fee of .15% per annum on the unused portion of the total \$45,000,000 commitment. Additionally, the Company is obligated to meet certain financial covenants, including the maintenance of specified levels of tangible net worth and net income.

THE BOSTON BEER COMPANY, INC.
NOTES TO CONSOLIDATED FINANCIAL STATEMENTS
(Continued)

#### G. RECENT ACCOUNTING STANDARDS:

In February, 1997, the Financial Accounting Standards Board issued Statement of Financial Accounting Standards No. 128, "Earnings Per Share" (SFAS 128), which modifies the way in which

EPS is calculated and disclosed. Currently, the Company discloses primary and fully diluted EPS. SFAS 128 requires the disclosure of basic and diluted EPS for financial statements issued for periods ending after December 15, 1997; the restatement of all prior period EPS data presented is also required upon adoption. Basic EPS excludes potentially dilutive securities and is computed by dividing net income (loss) available to common stockholders by the weighted-average number of common shares outstanding for the period. Diluted EPS, similar to fully diluted EPS, reflects the potential dilution that could occur if securities or other contracts to issue common stock were exercised or converted into common stock that then shared in the earnings of the entity. Early application of SFAS 128 is not permitted.

The following table summarizes the Company's EPS and weighted-average common and common equivalent shares outstanding as reported and on a pro forma basis as calculated under SFAS 128. The pro forma results for the diluted calculation do not differ materially from the fully diluted calculation; therefore, no pro forma information for the diluted calculation has been presented.

		Three Months E	Inded	
	March 29, 1997		March 30, 1996	
	As Reported Primary	Pro forma Basic	As Reported Primary	Pro forma Basic
EPS: Net income per share	\$ .08	\$ .08	\$ .13	\$ .13
Weighted-average common and common equivalent shares outstanding (in thousands)	20,313	20,133	20,366 ======	19,861 =====

THE BOSTON BEER COMPANY, INC.

# Item 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following is a discussion of the financial condition and results of operations of the Company for the three-month period ended March 29, 1997 as compared to the three-month period ended March 30, 1996. It should be read in conjunction with the "Consolidated Financial Statements" of the Company and related "Notes to the Financial Statements" included in this Form 10-Q. On March 1, 1997, the Company acquired all of the equipment and other brewery-related personal property from The Schoenling Brewing Company, located in Cincinnati, Ohio and leased the real-estate on which the brewery is situated.

#### RESULTS OF OPERATIONS

Three Months Ended March 29, 1997 compared to Three Months Ended March 30, 1996.

Sales volume increased by 2% from 276,000 barrels in the three months ended March 30, 1996 to 281,000 barrels in the three months ended March 29,1997. This increase was due to inclusion of 19,000 barrels, in 1997, from the Hudepohl-Schoenling brewery, which was acquired on March 1, 1997. Net sales decreased by 3% from \$43,129,000

in the three month period ended March 30, 1996 to \$41,855,000 in the three month period ended March 29, 1997. The net sales percentage decrease was primarily due to a shift in package mix (i.e. a higher percentage of kegs, which produce less revenue per barrel, versus cases than the Company has historically experienced). Towards the end of the quarter, the shift in package mix moved back toward the Company's normal package mix. It is probable that the sales volume during the second quarter of 1997 will be less than that of the second quarter of 1996.

Gross profit decreased by 6% from \$21,264,000 in the three months ended March 30, 1996 to \$19,948,000 in the three months ended March 29, 1997. Cost of sales increased to 52% of net sales in the three months ended March 29, 1997 from 51% in the three months ended March 30, 1996. This increase was primarily due to increases in raw materials costs due to product mix (i.e. higher percent of Seasonal beers in the three months ended March 29, 1997 as compared to the three months ended March 30, 1996), obsolescence expense, and in depreciation (primarily due to the newly purchased Sankey kegs), partially offset by a decrease in certain packaging material costs.

Advertising, promotional, and selling expenses in total increased by 4% from \$14,029,000 in the three months ended March 30, 1996 to \$14,558,000 in the three months ended March 29, 1997. This increase was primarily attributable to increased promotions, salaries and related personnel expenses due to additional sales force hires, partially offset by a rescheduling of certain point of sale programs, which may be a shift of expense to future quarters, and lower per case freight due to a shift in distribution patterns. As a percentage of net sales, total advertising, promotional, and selling expenses increased from 33% in the three months ended March 30, 1996 to 35% in the three months ended March 29, 1997.

General and administrative expenses decreased by 2% from \$2,983,000 in the three months ended March 30, 1996 to \$2,930,000 in the three months ended March 29, 1997. This decrease was primarily due to decreases in consulting fees and R&D material charges, offset somewhat by an increase in depreciation, primarily reflecting leasehold expansion and improvements as well as an increase in insurance. As a percentage of net sales, general and administrative expenses remained stable at 7% for the three months ended March 30, 1996 and 7% for the three months ended March 29, 1997.

#### THE BOSTON BEER COMPANY, INC.

# Item 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS (Continued)

Other income (expense) net, for the three months ended March 29, 1997, was \$350,000, representing a decrease of \$84,000 over other income (expense) net, for the three months ended March 30, 1996. The decrease in other income for the three months ended March 29, 1997 is due to a reduction in invested cash caused primarily by the purchase of the previously mentioned assets of The Schoenling Brewing Company as well as the interest expense related to the borrowings against the revolving line of credit, offset by the interest income earned on the remaining proceeds from the Company's November 1995 sales of shares of its Class A Common Stock.

Net income decreased by 40% to \$1,580,000 in the three months ended March 29, 1997 compared to \$2,640,000 in the three months ended March 30, 1996. The combined effective tax rate increased to 43.8% in the three months ended March 29, 1997 compared to 43.7% in the three months ended March 30, 1996, due primarily to higher projected income levels for 1997. It is possible that these projected income levels may

not be attained.

Liquidity and Capital Resources

During the first quarter of 1997, the Company recorded net income of \$1,580,000, while net cash used by operating activities was \$8,035,000. This \$9,615,000 difference is primarily due to increases in accounts receivable and inventory, and a reduction in accounts payable.

During the first quarter of 1997, the Company used \$6,869,000 in investing activities. This primarily reflects the addition of fixed assets of brewery and packaging equipment and the purchase of the equipment and other brewery-related personal property of The Schoenling Brewery.

During the first quarter of 1997, the Company borrowed against its revolving line of credit. As of March 29, 1997, net borrowings were \$9,871,000.

Effective on March 21, 1997, the Company negotiated an additional \$1,000,000 on the existing \$14,000,000 revolving line of credit, and secured an additional revolving line of credit of \$30,000,000 which will convert to a term loan on March 31, 1999.

The Company believes that working capital as of March 29, 1997 of \$41,269,000 (of which 86% is in cash and equivalents and short term investments) in conjunction with existing lines of credit should be sufficient to meet the Company's operating, capital, and debt service requirements during the remainder of 1997.

Recent Accounting Standards Pronouncements

In February, 1997, the Financial Accounting Standards Board issued Statement of Financial Accounting Standards No. 128, "Earnings Per Share" (SFAS 128), which modifies the way in which EPS is calculated and disclosed. Currently, the Company discloses primary and fully diluted EPS. SFAS 128 requires the disclosure of basic and diluted EPS for financial statements issued for periods ending after December 15, 1997; the restatement of all prior period EPS data presented is also required upon adoption. Basic EPS excludes potentially dilutive securities and is computed by dividing net income (loss) available to common stockholders by the weighted-average number of common shares outstanding for the period. Diluted EPS, similar to fully diluted EPS, reflects the potential dilution that could occur if securities or other contracts to issue common stock were exercised or converted into common stock that then shared in the earnings of the entity. Early application of SFAS 128 is not permitted.

THE BOSTON BEER COMPANY, INC.

Item 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS (Continued)

Recent Accounting Standards Pronouncements (continued)

For the three months ended March 29, 1997, the as reported Primary and the pro forma Basic earnings per share were \$.08 and \$.08, and the weighted average common and common equivalent shares outstanding, in thousands, were 20,313 and 20,133, respectively.

For the three months ended March 30, 1996, the as reported Primary and the pro forma Basic earnings per share were \$.13 and \$.13, and the weighted average common and common equivalent shares outstanding, in thousands, were 20,366 and 19,861, respectively.

For both periods presented, the pro forma results for the diluted calculation do not differ materially from the fully diluted calculation; therefore, no pro forma information for the diluted calculation has been presented.

Forward-Looking Statements

The above discussion includes forward-looking statements. Such forward-looking statements involve known and unknown risks, uncertainties, and other factors which may cause the actual results, performance or achievements of the Company to be materially different from any future results, performance, or achievements expressed or implied by such forward-looking statements. Such factors include, among others, the following: changes in consumer preferences; general economic and business conditions; increasing competition in the craft-brewed beer industry; success of operating initiatives; possible future increases in operating costs; advertising and promotional efforts; brand awareness; the existence or absence of adverse publicity; changes in business strategy; quality of management; availability, terms and deployment of capital; business abilities and judgment of personnel; availability of qualified personnel; labor and employee benefit costs; change in, or the failure to comply with, government regulations; and other factors.

THE BOSTON BEER COMPANY, INC.

# PART II. OTHER INFORMATION

#### Item 1. LEGAL PROCEEDINGS

In early 1996, Boston Brewing Company, Inc. ("Boston Brewing"), an affiliate of both Boston Beer Company Limited Partnership and The Boston Beer Company, Inc., had an action filed against it by its distributor, Premier Worldwide Beers PLC ("Premier"), such action having been filed in a court in England. Premier's action contains a claim to damages for alleged breach of a Distributorship Agreement between Boston Brewing and Premier. The action is being vigorously defended and at present is in the discovery stage.

The Company is party to certain claims and litigation in the ordinary course of business. The Company does not believe any of these proceedings will result, individually or in the aggregate, in a material adverse effect upon its financial condition or results of operations.

Item 2. CHANGES IN SECURITIES

Not Applicable

Item 3. DEFAULTS UPON SENIOR SECURITIES

Not Applicable

Item 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

Not Applicable

Item 5. OTHER INFORMATION

Not Applicable

Item 6. EXHIBITS AND REPORTS ON FORM 8-K

(a) Exhibits

The following is a list of exhibits filed as part of this report:

Exhibit No.	Title
3.1	Articles of Organization (incorporated by reference to Exhibit 3.2 to the Company's Registration Statement No. 33-96162).
3.2	By-Laws of the Company (incorporated by reference to Exhibit 3.2 to the Company's Registration Statement No. 33-96162).
3.3	Restated Articles of Organization of the Company (incorporated by reference to Exhibit 3.3 to the Company's Form 10-K filed on April 1, 1996).

THE BOSTON BEER COMPANY, INC.

# PART II. OTHER INFORMATION (continued)

Item 6. EXHIBITS AND REPORTS ON FORM 8-K (continued)

Exhibit No.	Title
3.4	Amended and Restated By-Laws of the Company (incorporated by reference to Exhibit 3.4 to the Company's Form 10-K filed on April 1, 1996).
4.1	Form of Class A Common Stock Certificate (incorporated by reference to Exhibit 4.1 to the Company's Registration Statement No. 33-96164).
10.1	Revolving Credit Agreement between Fleet Bank of Massachusetts, N.A. and Boston Beer Company Limited Partnership (the "Partnership"), dated as of May 2, 1995 (incorporated by reference to Exhibit 10.2 to the Company's Registration Statement No. 33-96162).
10.2	Loan Security and Trust Agreement, dated October 1, 1987, among Massachusetts Industrial Finance Agency, the Partnership and The First National Bank of Boston, as Trustee, as amended (incorporated by reference to Exhibit 10.2 to the Company's Registration Statement No. 33-96164).
10.3	Deferred Compensation Agreement between the Partnership and Alfred W. Rossow, Jr., effective December 1, 1992 (incorporated by reference to Exhibit 10.3 to the Company's Registration Statement No. 33-96162).
10.4	The Boston Beer Company, Inc. Employee Equity Incentive Plan, as adopted effective November 20, 1995 and amended effective February 23, 1996 (incorporated by reference to Exhibit 4.1 to the Company's Registration Statement No. 333-1798).
10.5	Form of Employment Agreement between the Partnership and employees (incorporated by reference to Exhibit 10.5 to the Company's Registration Statement No. 33-96162).

- 10.6 Services Agreement between The Boston Beer Company, Inc. and Chemical Mellon Shareholder Services, dated as of October 27, 1995 (incorporated by reference to the Company's Form 10-K, filed on April 1, 1996).
- 10.7 Form of Indemnification Agreement between the Partnership and certain employees and Advisory Committee members (incorporated by reference to Exhibit 10.7 to the Company's Registration Statement No. 33-96162).

THE BOSTON BEER COMPANY, INC.

#### PART II. OTHER INFORMATION (continued)

#### Item

OTHER INFORMA	ATION (continued)
6. EXHIBITS AND	REPORTS ON FORM 8-K (continued)
Exhibit No.	Title
10.8	Stockholder Rights Agreement, dated as of December, 1995, among The Boston Beer Company, Inc. and the initial Stockholders (incorporated by reference to the Company's Form 10-K, filed on April 1, 1996).
+10.9	Agreement between Boston Brewing Company, Inc. and The Stroh Brewery Company, dated as of January 31, 1994 (incorporated by reference to Exhibit 10.9 to the Company's Registration Statement No. 33-96164).
+10.10	Agreement between Boston Brewing Company, Inc. and the Genesee Brewing Company, dated as of July 25, 1995 (incorporated by reference to Exhibit 10.10 to the Company's Registration Statement No. 33-96164).
+10.11	Amended and Restated Agreement between Pittsburgh Brewing Company and Boston Brewing Company, Inc. dated as of February 28, 1989 (incorporated by reference to Exhibit 10.11 to the Company's Registration Statement No. 33-96164).
10.12	Amendment to Amended and Restated Agreement between Pittsburgh Brewing Company, Boston Brewing Company, Inc., and G. Heileman Brewing Company, Inc., dated December 13, 1989 (incorporated by reference to Exhibit 10.13 to the Company's Registration Statement No. 33-96162).
+10.13	Second Amendment to Amended and Restated Agreement between Pittsburgh Brewing Company and Boston Brewing Company, Inc. dated as of August 3, 1992 (incorporated by reference to Exhibit 10.13 to the Company's Registration Statement No. 33-96164).
+10.14	Third Amendment to Amended and Restated Agreement between Pittsburgh Brewing Company and Boston Brewing Company, Inc. dated December 1, 1994 (incorporated by reference to Exhibit 10.14 to the Company's Registration Statement

No. 33-96164).

10.15 Fourth Amendment to Amended and Restated Agreement between Pittsburgh Brewing Company and Boston Brewing Company, Inc. dated as of April 7, 1995 (incorporated by reference to Exhibit 10.16 to the Company's Registration Statement No. 33-96162).

THE BOSTON BEER COMPANY, INC.

# PART II. OTHER INFORMATION (continued)

# Item 6. EXHIBITS AND REPORTS ON FORM 8-K (continued)

Exhibit	No.	Title
+10.16		Letter Agreement between Boston Beer Company Limited Partnership and Joseph E. Seagram & Sons, Inc.(incorporated by reference to Exhibit 10.17 to the Company's Registration Statement No. 33-96162).
10.17		Services Agreement and Fee Schedule of Mellon Bank, N.A. Escrow Agent Services for The Boston Beer Company, Inc. dated as of October 27, 1995).
10.18		Amendment to Revolving Credit Agreement between Fleet Bank of Massachusetts, N.A. and the Partnership (incorporated by reference to Exhibit 10.17 to the Company's Registration Statement No. 33-96164).
10.19		1996 Stock Option Plan for Non-Employee Directors (incorporated by reference to the Company's Form 10-K, filed on March 28, 1997).
+10.20		Production Agreement between The Stroh Brewery Company and Boston Beer Company Limited Partnership, dated January 14, 1997 (incorporated by reference to the Company's Form 10-K, filed on March 28, 1997).
+10.21		Letter Agreement between The Stroh Brewery Company and Boston Beer Company Limited Partnership, dated January 14, 1997 (incorporated by reference to the Company's Form 10-K, filed on March 28, 1997).
+10.22		Agreement between Boston Beer Company Limited Partnership and The Schoenling Brewing Company, dated May 22, 1996 (incorporated by reference to the Company's Form 10-K, filed on March 28, 1997).
*10.23		Revolving Credit Agreement between Fleet Bank of Massachusetts, N.A. and The Boston Beer Company, Inc., dated as of March 21, 1997.
*11		Schedule of Computation of Net Earnings Per Share.
21.1		List of subsidiaries of The Boston Beer Company, Inc. (incorporated by reference to the Company's Form 10-K, filed on March 28, 1997).

# THE BOSTON BEER COMPANY, INC.

#### PART II. OTHER INFORMATION (continued)

#### Item 6. EXHIBITS AND REPORTS ON FORM 8-K (continued)

Exhibit No. Title

- \* Filed with this report.
- + Portions of this Exhibit have been omitted pursuant to an application for an order declaring confidential treatment filed with the Securities and Exchange Commission.
- (b) Reports on Form 8-K.

Not Applicable.

THE BOSTON BEER COMPANY, INC.

# EXHIBIT INDEX

Exhibit Number	Description of Exhibits
10.23	Revolving Credit Agreement between Fleet Bank of Massachusetts, N.A. and The Boston Beer Company, Inc., dated as of March 21, 1997.
11	Schedule of Computation of Net Earnings Per Share.

# SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this Form 10-Q to be signed on its behalf by the undersigned thereunto duly authorized.

Date: May 12, 1997 By: /s/ C. JAMES KOCH

C. James Koch
President, Chief Executive
Officer, Clerk and Director
(principal executive officer)

Date: May 12, 1997 By: /s/ ALFRED W. ROSSOW, JR.

Alfred W. Rossow, Jr.
Executive Vice President,
Chief Financial Officer
(principal financial and
accounting officer)
Treasurer, and Director

# Exhibit 11.

# THE BOSTON BEER COMPANY, INC. STATEMENT REGARDING COMPUTATION OF NET EARNINGS PER SHARE (Unaudited)

(in thousands, except per share data)

	Quarter Ended		
	March 29, 1997	March 30, 1996	
Weighted average number of common shares outstanding	20,132,941	19,860,779	
Add:Common equivalent shares representing shares issuable upon conversion of stock options (using the treasury stock method)	180,226	505 <b>,</b> 064	
Weighted average number of common and common equivalent shares	20,313,167	20,365,843	
Net income	\$ 1,580 ======	\$ 2,640 ======	
Primary and fully diluted earnings per share	\$ .08 ======	\$ .13 ======	

The accompanying notes are an integral part of the financial statements.

# <ARTICLE> 5

<LEGEND>

THIS SCHEDULE CONTAINS SUMMARY FINANCIAL INFORMATION EXTRACTED FROM THE BOSTON BEER COMPANY, INC.'S CONSOLIDATED BALANCE SHEET AND CONSOLIDATED STATEMENTS OF INCOME AND IS QUALIFIED IN ITS ENTIRETY BY REFERENCE TO SUCH FINANCIAL STATEMENTS.

</LEGEND>

<MULTIPLIER> 1,000
<CURRENCY> US DOLLAR

<period-type></period-type>	3-MOS	
<fiscal-year-end></fiscal-year-end>		DEC-27-1997
<period-start></period-start>		DEC-29-1996
<period-end></period-end>		MAR-29-1997
<exchange-rate></exchange-rate>		1.000
<cash></cash>		32
<securities></securities>		35536
<receivables></receivables>		21089
<allowances></allowances>		1953
<inventory></inventory>		14254
<current-assets></current-assets>		75397
<pp&e></pp&e>		29069
<depreciation></depreciation>		7378
<total-assets></total-assets>		101437
<current-liabilities></current-liabilities>		34058
<bonds></bonds>		1800
<preferred-mandatory></preferred-mandatory>		0
<preferred></preferred>		0
<common></common>		201
<other-se></other-se>		65378
<total-liability-and-equity></total-liability-and-equity>		101437
<sales></sales>		46799
<total-revenues></total-revenues>		41855
<cgs></cgs>		21907
<total-costs></total-costs>		39395
<other-expenses></other-expenses>		0
<loss-provision></loss-provision>		0
<interest-expense></interest-expense>		108
<income-pretax></income-pretax>		2810
<income-tax></income-tax>		1230
<pre><income-continuing></income-continuing></pre>		1580
<discontinued></discontinued>		0
<extraordinary></extraordinary>		0
<changes></changes>		0
<net-income></net-income>		1580
<eps-primary></eps-primary>		.08
<eps-diluted></eps-diluted>		.08

# CREDIT AGREEMENT

among

# FLEET NATIONAL BANK,

# BOSTON BEER COMPANY LIMITED PARTNERSHIP

and

THE BOSTON BEER COMPANY, INC.

Dated as of March 21, 1997

# TABLE OF CONTENTS

Sect	ion	Page
1 -	DESCRIPTION OF CREDIT	1
1.1	The Loans	1
1.2	The Notes	2
1.3	Interest; Default Rate	2
1.4	Notice and Manner of Borrowing or Conversion of Loans	2
1.5	Prepayments	3
1.6	Fees	3
1.7	Capital Requirements	3
1.8	Method of Payment	3
1.9	Computation of Interest and Fees	3
1.10	Use of Proceeds	3

	4	
2.1 Conditions Precedent to Initial Loan	4	
2.2 Conditions Precedent to all Loans	4	
3 - REPRESENTATIONS AND WARRANTIES	4	
3.1 Organization and Qualification	4	
3.2 Authority	5	
3.3 Consents; Validity	5	
3.4 Title to Properties; Absence of Encumbrances	5	
3.5 Financial Statements	5	
3.6 Taxes	6	
3.7 Litigation	6	
3.8 Compliance with Laws and Agreements	6	
3.9 No Insolvency	6	
3.10 Full Disclosure	6	
3.11 Margin Stock	7	
3.12 Patents, Trademarks, Etc.	7	
3.13 ERISA	7	
3.14 Environmental Laws	7	
3.15 Ownership	8	
4 - AFFIRMATIVE COVENANTS	9	
4.1 Financial Reporting	9	
4.2 Conduct of Business	10	
4.3 Maintenance and Insurance	10	
4.4 Taxes and Accounts Payable	10	
4.5 Maintenance of Books and Records	10	
4.6 Environmental Laws	11	
4.7 Further Assurances	11	
4.8 Principal Depository	11	
110 Principal Popolicoli		
Section	Page	
5 - FINANCIAL COVENANTS	11	
5 - FINANCIAL COVENANIS	1.1	
	11	
5.1 Covenants		
5.1 Covenants 5.2 Definitions	11	
<ul><li>5.2 Definitions</li><li>6 - NEGATIVE COVENANTS</li></ul>	11	
<ul><li>5.2 Definitions</li><li>6 - NEGATIVE COVENANTS</li><li>6.1 Indebtedness, Guarantees</li></ul>	11 12 12	
<ul><li>5.2 Definitions</li><li>6 - NEGATIVE COVENANTS</li><li>6.1 Indebtedness, Guarantees</li><li>6.2 Encumbrances</li></ul>	11 12 12 13	
<ul> <li>5.2 Definitions</li> <li>6 - NEGATIVE COVENANTS</li> <li>6.1 Indebtedness, Guarantees</li> <li>6.2 Encumbrances</li> <li>6.3 Sale and Leaseback</li> </ul>	11 12 12 13 13	
<ul> <li>5.2 Definitions</li> <li>6 - NEGATIVE COVENANTS</li> <li>6.1 Indebtedness, Guarantees</li> <li>6.2 Encumbrances</li> <li>6.3 Sale and Leaseback</li> <li>6.4 Merger; Consolidation; Disposition of Assets, Etc.</li> </ul>	11 12 12 13 13	
<ul> <li>5.2 Definitions</li> <li>6 - NEGATIVE COVENANTS</li> <li>6.1 Indebtedness, Guarantees</li> <li>6.2 Encumbrances</li> <li>6.3 Sale and Leaseback</li> <li>6.4 Merger; Consolidation; Disposition of Assets, Etc.</li> <li>6.5 Investments</li> </ul>	11 12 12 13 13 14 14	
5.2 Definitions  6 - NEGATIVE COVENANTS  6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates	11 12 12 13 13 14 14	
<ul> <li>5.2 Definitions</li> <li>6 - NEGATIVE COVENANTS</li> <li>6.1 Indebtedness, Guarantees</li> <li>6.2 Encumbrances</li> <li>6.3 Sale and Leaseback</li> <li>6.4 Merger; Consolidation; Disposition of Assets, Etc.</li> <li>6.5 Investments</li> </ul>	11 12 12 13 13 14 14	
5.2 Definitions  6 - NEGATIVE COVENANTS  6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates	11 12 12 13 13 14 14	
5.2 Definitions  6 - NEGATIVE COVENANTS  6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates 6.7 Distributions	11 12 12 13 13 14 14 14	
<pre>5.2 Definitions 6 - NEGATIVE COVENANTS 6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates 6.7 Distributions 7 - DEFAULTS 7.1 Events of Default</pre>	11 12 12 13 13 14 14 14	
5.2 Definitions  6 - NEGATIVE COVENANTS  6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates 6.7 Distributions  7 - DEFAULTS	11 12 12 13 13 14 14 14 14	
<pre>5.2 Definitions 6 - NEGATIVE COVENANTS 6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates 6.7 Distributions 7 - DEFAULTS 7.1 Events of Default</pre>	11 12 12 13 13 14 14 14 14	
<pre>5.2 Definitions 6 - NEGATIVE COVENANTS 6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates 6.7 Distributions 7 - DEFAULTS 7.1 Events of Default 7.2 Remedies 8 - GENERAL PROVISIONS</pre>	11 12 12 13 13 14 14 14 14 14 16	
5.2 Definitions  6 - NEGATIVE COVENANTS  6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates 6.7 Distributions  7 - DEFAULTS  7.1 Events of Default 7.2 Remedies  8 - GENERAL PROVISIONS  8.1 Notices	11 12 12 13 13 14 14 14 14 16 16	
5.2 Definitions  6 - NEGATIVE COVENANTS  6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates 6.7 Distributions  7 - DEFAULTS  7.1 Events of Default 7.2 Remedies  8 - GENERAL PROVISIONS  8.1 Notices 8.2 Expenses	11 12 12 13 13 14 14 14 14 16 16 16	
5.2 Definitions  6 - NEGATIVE COVENANTS  6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates 6.7 Distributions  7 - DEFAULTS  7.1 Events of Default 7.2 Remedies  8 - GENERAL PROVISIONS  8.1 Notices 8.2 Expenses 8.3 Set-Off	11 12 12 13 13 14 14 14 14 16 16 16 16 17 17	
5.2 Definitions 6 - NEGATIVE COVENANTS 6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates 6.7 Distributions 7 - DEFAULTS 7.1 Events of Default 7.2 Remedies 8 - GENERAL PROVISIONS 8.1 Notices 8.2 Expenses 8.3 Set-Off 8.4 No Waivers; Term	11 12 12 13 13 14 14 14 14 16 16 16 16 17 17	
5.2 Definitions 6 - NEGATIVE COVENANTS 6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates 6.7 Distributions 7 - DEFAULTS 7.1 Events of Default 7.2 Remedies 8 - GENERAL PROVISIONS 8.1 Notices 8.2 Expenses 8.3 Set-Off 8.4 No Waivers; Term 8.5 Governing Law	11 12 12 13 13 14 14 14 14 16 16 16 16 17 17 17	
5.2 Definitions 6 - NEGATIVE COVENANTS 6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates 6.7 Distributions 7 - DEFAULTS 7.1 Events of Default 7.2 Remedies 8 - GENERAL PROVISIONS 8.1 Notices 8.2 Expenses 8.3 Set-Off 8.4 No Waivers; Term 8.5 Governing Law 8.6 Amendments	11 12 12 13 13 14 14 14 14 16 16 16 16 17 17 17	
5.2 Definitions 6 - NEGATIVE COVENANTS 6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates 6.7 Distributions 7 - DEFAULTS 7.1 Events of Default 7.2 Remedies 8 - GENERAL PROVISIONS 8.1 Notices 8.2 Expenses 8.3 Set-Off 8.4 No Waivers; Term 8.5 Governing Law 8.6 Amendments 8.7 Binding Effect of Agreement	11 12 12 13 13 14 14 14 14 14 16 16 16 17 17 17 17	
5.2 Definitions 6 - NEGATIVE COVENANTS 6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates 6.7 Distributions 7 - DEFAULTS 7.1 Events of Default 7.2 Remedies 8 - GENERAL PROVISIONS 8.1 Notices 8.2 Expenses 8.3 Set-Off 8.4 No Waivers; Term 8.5 Governing Law 8.6 Amendments 8.7 Binding Effect of Agreement 8.8 Entire Agreement	11 12 12 13 13 14 14 14 14 14 16 16 16 17 17 17 17 17 17 18	
5.2 Definitions 6 - NEGATIVE COVENANTS 6.1 Indebtedness, Guarantees 6.2 Encumbrances 6.3 Sale and Leaseback 6.4 Merger; Consolidation; Disposition of Assets, Etc. 6.5 Investments 6.6 Transactions with Affiliates 6.7 Distributions 7 - DEFAULTS 7.1 Events of Default 7.2 Remedies 8 - GENERAL PROVISIONS 8.1 Notices 8.2 Expenses 8.3 Set-Off 8.4 No Waivers; Term 8.5 Governing Law 8.6 Amendments 8.7 Binding Effect of Agreement	11 12 12 13 13 14 14 14 14 14 16 16 16 17 17 17 17	

#### SCHEDULES

Schedule A General Defined Terms
Schedule B Libor Terms
Schedule CRepresentations and Warranties

#### EXHIBITS

Exhibit 1.2(a) Form of Revolving Credit Note

Exhibit 1.2(b) Form of Revolving Credit and Term Note

Exhibit 1.4 Notice of Borrowing or Conversion

Exhibit 2.1 Closing Agenda

Exhibit 3.7 Litigation

Exhibit 3.14Environmental Matters

Exhibit 3.15 Ownership

Exhibit 4.1 Form of Officer's Compliance Report

#### CREDIT AGREEMENT

THIS AGREEMENT is made as of March 21, 1997 by and among BOSTON BEER COMPANY LIMITED PARTNERSHIP, a Massachusetts limited partnership (the "Partnership") and THE BOSTON BEER COMPANY, INC. (the "Corporation") (collectively, the "Borrowers" and individually, a "Borrower"); and FLEET NATIONAL BANK, a national banking association (the "Bank"). Capitalized terms used herein are defined in Section 5 and Schedules A and B hereto.

- A. The Partnership and Fleet Bank of Massachusetts, N.A., predecessor in interest to the Bank, are parties to a Revolving Credit Agreement dated as of May 2, 1995, as amended (the "Original Agreement"), pursuant to which the Partnership issued its Revolving Note (the "Original Revolving Note") to evidence a revolving line of credit (the "Existing Line of Credit") in the principal amount of \$14,000,000,\$-0- of which is outstanding.
- B. The Borrowers desire to increase their Existing Line of Credit to \$15,000,000 and desire to establish an additional revolving line of credit in the amount of \$30,000,000 which would convert to a term loan on the Conversion Date, as provided herein.
- C. The Bank is willing to provide such additional funds to the Borrowers hereunder, subject to the terms and conditions set forth herein. This Agreement amends and restates the Original Agreement.

NOW, THEREFORE, in consideration of the premises and the mutual agreements herein contained, the parties hereto hereby agree as follows:

### 1. DESCRIPTION OF CREDIT.

#### 1.1 The Loans.

(a) Subject to the terms and conditions of this Agreement, the Bank will make loans (the "Revolving Loans A") jointly to the Borrowers from time to time until the close of business on March 31, 1999 (the "Expiration Date"), in such sums as the Borrowers may request; provided, however, the aggregate principal amount of Revolving Loans A at any time outstanding hereunder shall not exceed \$15,000,000 (the "Revolver A Commitment"). The Borrowers may borrow, prepay and reborrow from the date hereof to the Expiration Date, the full amount of the Revolver A Commitment or

any lesser sum; provided that Libor Loans shall be in an amount of at least \$100,000 and an integral multiple of \$100,000. All of the outstanding principal on the Revolving Loans A shall be due and payable in full on the Expiration Date.

- (b) Subject to the terms and conditions of this Agreement, the Bank will make loans (the "Revolving Loans") jointly to the Borrowers from time to time until the close of business on March 31, 1999 (the "Conversion Date"), in such sums as the Borrowers may request; provided, however, the aggregate principal amount of all Revolving Loans B at any time outstanding hereunder shall not exceed \$30,000,000 (the "Revolver B Commitment"). The Borrowers may borrow, prepay and reborrow from the date hereof to the Conversion Date, the full amount of the Revolver B Commitment or any lesser sum; provided that Libor Loans shall be in an amount of at least \$100,000 and an integral multiple of \$100,000.
- (c) Subject to the terms and conditions of this Agreement, the Bank and the Borrowers agree that the Revolving Loans B shall convert to a term loan (the "Term Loan") on the Conversion Date in the principal amount (the "Term Loan Principal") of the then outstanding principal amount of the Revolving Loans B, but in any event not in excess of the Revolver B Commitment. Any funds repaid on or after the Conversion Date may not be readvanced thereafter. In the event the terms and conditions set forth in Section 2.2 for the Term Loan are not met in full as of the Conversion Date, the outstanding principal balance of the Revolving Loans B shall be due and payable in full on the Conversion Date. The Term Loan Principal shall be payable without setoff, deduction or counterclaim in twenty equal consecutive quarterly installments on the last day of each fiscal quarter, commencing March 31, 1999 and continuing until December 31, 2003, when all outstanding principal and accrued interest thereon shall be due and payable in full without setoff, deduction or counterclaim (the "Maturity Date").
- 1.2 The Notes. The Revolving Loans A shall be evidenced by the Borrower's Revolving Credit Note in the form of Exhibit 1.2 (a) hereto and the Revolving Loans B shall be evidenced by the Borrowers' Revolving Credit and Term Note, in the form of Exhibit 1.2(b) hereto, each payable to the order of the Bank (collectively, the "Notes"), said Revolving Credit Note amends and restates the Original Revolving Note. The Notes are incorporated herein by reference and made a part hereof.

# 1.3 Interest; Default Rate.

- (a) Subject to this Agreement, the Borrowers may elect an interest rate for each Revolving Loan A and Revolving Loan B based on either (i) the Alternative Prime Rate or (ii) the applicable Adjusted Libor Rate (as defined on Schedule B hereto) plus .25%. Subject to the terms and conditions of this Agreement, the Borrowers may elect an interest rate for the Term Loan based on either (i) the Alternative Prime Rate or (ii) the applicable Adjusted Libor Rate plus .50%. Each Prime Rate Loan shall bear interest on the outstanding principal amount thereof at a rate per annum equal to the Prime Rate (which rate shall change contemporaneously with any change in the Prime Rate), payable on the last day of each fiscal quarter, commencing on March 31, 1997, and when such Prime Rate Loan is due (whether at maturity, by reason of acceleration or otherwise). Libor Loans shall bear interest, and otherwise be governed, in accordance with Schedule B (the "Libor Terms").
- (b) Overdue principal (whether at maturity, by reason of acceleration or otherwise) and, to the extent permitted by applicable law, overdue interest and fees or any other amounts payable hereunder or under the Notes, shall bear interest from

and including the due date until paid, payable on demand, at a rate per annum equal to 3% above the rate then applicable to Prime Rate Loans. In addition, if the entire amount of any required principal and/or interest is not paid in full within ten days after the same is due, the Borrowers shall pay to the Bank a late fee equal to 5% of the required payment.

- 1.4 Notice and Manner of Borrowing or Conversion of Loans.
- (a) Whenever the Borrowers desire to obtain or continue a Loan hereunder or convert an outstanding Loan to a Loan of a different type, it shall notify the Bank (which notice shall be irrevocable) by telex, telegraph or telephone received no later than 10:30 a.m. Boston time on the date (i) of the requested action, with respect to a Loan to be made, continued or converted to a Prime Rate Loan, and (ii) two Business Days before the requested action, with respect to a Loan to be made, continued as or converted to a Libor Loan. Such notice shall specify (A) the effective date and amount of each Loan or portion thereof to be made, continued or converted, (B) the interest rate option to be applicable thereto, and (C) the duration of the Libor Period, if applicable. Each such notification (a "Notice of Borrowing or Conversion") shall be immediately followed by a written confirmation in the form attached hereto as Exhibit 1.4.
- (b) Subject to the Libor Terms, the Borrowers may convert all or any part (in integral multiples of \$100,000) of any outstanding Loan into a Libor Loan or a Prime Rate Loan, as the case may be, in the same aggregate principal amount, on any Business Day (which, in the case of a conversion of an existing

Libor Loan, must be on the last day of the Libor Period

applicable to such existing Libor Loan).

- 1.5 Prepayments. Libor Loans may be prepaid, without premium or penalty (but subject to the Libor Terms), on the last day of the applicable Libor Period, upon two Business Days' notice. Prime Rate Loans may be prepaid at any time, without premium or penalty, on same day notice if such notice is received prior to 10:30 a.m. on the date of such prepayment. Any interest accrued on the amounts so prepaid to the date of such payment must be paid at the time of any such payment.
- 1.6 Fees. The Borrowers shall pay to the Bank during the Revolving Credit Period a non-refundable commitment fee computed at the rate of .15% per annum on the average daily unused portion of the Commitments, payable quarterly in arrears, on the last day of March, June, September and December of each year and on the last day of the Revolving Credit Period.
- 1.7 Capital Requirements. The Bank shall notify the Borrowers if after the date hereof the Bank determines that (a) the adoption after the date hereof of or change after the date hereof in any law, rule, regulation or guideline regarding capital requirements for banks or bank holding companies, or any change after the date hereof in the interpretation or application thereof by any governmental authority charged with the administration thereof (to the extent such adoption or change occurs after the date hereof), or (b) compliance by the Bank or its parent bank holding company with any quideline, request or directive taking effect after the date hereof of any such entity regarding capital adequacy (whether or not having the force of law), has the effect of reducing the return on the Bank's or such holding company's capital as a consequence of the Bank's commitment to make Loans hereunder to a level below that which the Bank or such holding company could have achieved but for such adoption, change or compliance (taking into consideration the Bank's or such holding company's then existing policies with

respect to capital adequacy and assuming the full utilization of such entity's capital) by any amount the Bank considers to be material. The Borrowers agree to pay to the Bank the actual amount of such reduction of return as and when such reduction is determined, when the Bank presents a statement of the amount and setting forth the Bank's calculation thereof. Such statement shall be deemed true and correct, absent manifest error.

- 1.8 Method of Payment. All payments and prepayments of principal and all payments of interest and fees shall be made by the Borrowers to the Bank at its head office in immediately available funds, on or before 11:00 a.m. (Boston time) on the due date thereof, free and clear of, and without set-off, deduction, withholding or counterclaim for, any taxes or other payments.
- 1.9 Computation of Interest and Fees. All interest and fees payable hereunder shall be computed daily on the basis of a 360-day year, counting the actual number of days elapsed. If the due date for any payment of principal is extended by operation of law, interest shall be payable for such extended time. If any payment required by this Agreement becomes due on a day that is not a Business Day, such payment may be made on the next succeeding Business Day (subject to the definition of Libor Period set forth in the Libor Terms), and such extension shall be included in computing interest in connection with such payment.
- 1.10 Use of Proceeds. The proceeds of the Revolving Loans A and Revolving Loans B shall be used (a) for the working capital requirements of the Corporation, (b) for fixed asset purchases by the Corporation made in the ordinary course of business and not pursuant to any business acquisition, and (c) up to \$5,000,000 in the aggregate for Permitted Acquisitions by the Corporation ("Acquisition Loans").

### 2. CONDITIONS OF LOANS.

- 2.1 Conditions Precedent to Initial Loan. The obligation of the Bank to make its initial Loan is subject to the conditions precedent that the Bank shall have received, in form and substance reasonably satisfactory to the Bank and its counsel, (a) all of the agreements, instruments and certificates described on the Closing Agenda attached as Exhibit 2.1, duly executed and delivered, together with all other documents and materials referred to in such Closing Agenda, and (b) satisfactory evidence of the completion of all actions referred to in such Closing Agenda.
- 2.2 Conditions Precedent to all Loans. The obligation of the Bank to make each Loan, including the initial Loan and the Term Loan, or to continue or convert Loans to Libor Loans, is further subject to the following conditions: (a) the Bank shall have received a properly completed Notice of Borrowing or Conversion, (b) the representations and warranties contained in Section 3 shall be true and accurate in all material respects on and as of the date of such Notice of Borrowing or Conversion and on the effective date of the making, continuation or conversion of each Loan as though made at and as of each such date (except to the extent that such representations and warranties expressly relate to an earlier date), (c) no Default shall have occurred and be continuing, or would result from such Loan, and (d) no change shall have occurred in any law or regulation or interpretation thereof making it illegal or against the policy of any governmental authority for the Bank to make Loans hereunder. Each request for the making, continuation or conversion of a Loan shall be deemed a representation and warranty to such effect by the Borrowers as of the date of the making, continuation or conversion of such Loan.

#### 3. REPRESENTATIONS AND WARRANTIES.

To induce the Bank to enter into this Agreement and to make Loans, each of the Borrowers makes the representations and warranties set forth below, each of which representations and warranties reflect the execution hereof and the initial Loan hereunder, and shall survive the delivery of the Notes and the making of all Loans:

#### 3.1 Organization and Qualification.

- (a) The Partnership (i) is limited partnership duly organized, validly existing and in good standing under the laws of the Commonwealth of Massachusetts, (ii) has all requisite partnership power to own its property and conduct its business as now conducted and as presently contemplated, (iii) is duly qualified and in good standing as a foreign limited partnership and is duly authorized to do business in each jurisdiction where the nature of its properties or business requires such qualification, and (iv) has no Subsidiaries, except SBCC Company, Inc., a Delaware corporation, Oregon Beer and Brewing Co., Inc. I, an Oregon corporation, and Samuel Adams Brewery Company, Ltd., an Ohio limited liability company. SBCC Company, Inc. is and will remain an inactive corporation; Oregon Beer and Brewing Co., Inc. I is an active operating corporation but not a Material Subsidiary; and Samuel Adams Brewery Company, Ltd., organized on January 14, 1997, became a Material Subsidiary and an active operating company on March 1, 1997, the effective date of its acquisition of brewery assets located in Cincinnati, Ohio, from The Schoenling Brewing Company. This representation shall be updated from time to time by the Borrowers when new Subsidiaries are formed in accordance with this Agreement.
- (b) The Corporation (i) is a Massachusetts corporation duly organized, validly existing and in good standing under the laws of the Commonwealth of Massachusetts, (ii) has all requisite corporate power to own its property and conduct its business as now conducted and as presently contemplated, (iii) is duly qualified and in good standing as a foreign corporation and is duly authorized to do business in each jurisdiction where the nature of its property or business requires such qualification, and (iv)has the Subsidiaries identified on Exhibit 3.15. Each of such Subsidiaries is engaging and will engage in no other business other than holding limited partnership interests in the Partnership.
- (c) Each of the Subsidiaries (i) is a corporation duly organized, validly existing and in good standing under its laws of its state of incorporation, (ii) has all requisite corporate power to own its own property and conduct its business as now conducted and is presently contemplated, (iii) is duly qualified and in good standing as a foreign corporation and is duly authorized to do business in each jurisdiction where the nature of its property or business requires such qualification, and (iv) has no Subsidiaries.
- 3.2 Authority. The execution, delivery and performance of this Agreement, the Notes, the guarantees of any Material Subsidiaries and all related agreements, instruments and certificates (all of the foregoing collectively referred to as the "Loan Documents") and the transactions contemplated hereby are within the partnership and corporate power and authority of the Borrowers and the Subsidiaries who are named as parties thereto, and have been authorized by the Corporation's Board of Directors for the Corporation itself and as sole general partner of the Partnership and by the Subsidiaries' Board of Directors, as applicable, and all other necessary proceedings, and do not and will not (a) violate the partnership agreement, certificate

of limited partnership, charter or bylaws of either Borrower or any Subsidiary or any law, rule, order or regulation applicable to any of them, (b) contravene any provision of, or constitute (with due notice or lapse of time or both), a default under, any other agreement, instrument, order or undertaking binding on either Borrower or any Subsidiary, or (c) result in the creation or imposition of any Encumbrance on any of the properties, assets or rights of either Borrower or any Subsidiary.

- 3.3 Consents; Validity. The execution, delivery and performance by each Borrower and each Subsidiary, if applicable, of the Loan Documents and the transactions contemplated therein do not require any approval or consent of, or filing or registration with, any governmental or other agency or authority or any other party. The Loan Documents are the legal, valid and binding obligations of each Borrower and each Subsidiary who are named as parties thereto, enforceable in accordance with their respective terms except as limited by bankruptcy, insolvency, reorganization, moratorium or other laws affecting the enforcement of creditors' rights generally.
- 3.4 Title to Properties; Absence of Encumbrances. Each Borrower and each Subsidiary has good title to all of the properties, assets and rights purported to be owned by it, including without limitation, those reflected in the financial statements referred to in Section 3.5 (except inventory and equipment disposed of in the ordinary course of business since the date thereof), free from all Encumbrances, except Permitted Encumbrances.
- 3.5 Financial Statements. The Corporation has furnished the Bank its balance sheet as of December 31, 1995 and its statements of operations and cash flows for the fiscal year then ended, and related footnotes, audited and certified by Coopers & Lybrand. The Partnership has furnished the Bank with its balance sheet as of December 31, 1995 and its statement of operations and cash flow for the fiscal year then ended, and related footnotes, audited and certified by Coopers and Lybrand. Each of the Corporation and the Partnership has furnished the Bank with its balance sheet as of September 30, 1996 and its statement of operations and cash flow for the 9-month period then ended, and related footnotes, prepared by the chief financial officer of the Corporation and the Partnership. All such financial statements were prepared in accordance with generally accepted accounting principles (subject to normal year-end adjustments in the ordinary course of business with respect to such nine-month interim statements) applied on a consistent basis throughout the periods specified and present fairly the financial position of each Borrower and their Subsidiaries as of such dates and the results of the operations of each Borrower and their Subsidiaries for such periods. There are no liabilities or commitments, contingent or otherwise, not disclosed in such financial statements that involve a material amount. Since the date of the most recent audited financial statements delivered to the Bank, except as otherwise disclosed in writing prior to the date hereof to the Bank with respect to certain unrealized loss in investment securities, there have been no changes in the assets, liabilities, financial condition or business of either Borrower or any Subsidiary, other than changes in the ordinary course of business, the effect of which has not, in the aggregate, been materially adverse to either Borrower or any Subsidiary or its assets and properties.
- 3.6 Taxes. Each Borrower and each Subsidiary has filed all federal, state and other material tax returns required to be filed by it, if any, and has paid or made adequate provision for the payment of all federal, state and other taxes, charges and assessments, if any.

- 3.7 Litigation. Except as disclosed on Exhibit 3.7, there is no litigation, arbitration, proceeding or investigation pending, or, to the knowledge of any of the Borrowers' officers or partners, threatened, against or affecting either Borrower or any Subsidiary that, if adversely determined, would reasonably be likely (a) to result in a material judgment not fully covered by insurance or a forfeiture of all or any material part of the property of either Borrower or any Subsidiary, or (b) otherwise to have a material adverse effect on the assets, business, prospects or condition, financial or otherwise, of either Borrower or any Subsidiary.
- 3.8 Compliance with Laws and Agreements. Neither Borrower nor any Subsidiary is a party to any agreement or instrument or subject to any corporate or partnership restriction materially adversely affecting its assets, business, prospects or condition, financial or otherwise. Neither Borrower nor any Subsidiary is in (a) violation or alleged violation of any provision of law or regulation, including without limitation any ERISA provision, any SEC law, rule or regulation or any Environmental Law, the violation of which would reasonably be likely to have a material adverse effect upon either Borrower or any Subsidiary, or any order, judgment or decree of any court or other agency of government or (b) default in the performance, observance or fulfillment of any of the obligations, covenants or conditions contained in any material agreement or instrument to which it is a party. Neither Borrower nor any Subsidiary is subject to regulation under the Investment Company Act of 1940, as amended.
- 3.9 No Insolvency. Neither Borrower nor any Subsidiary is insolvent or unable to pay its debts as they become due; the Loans made or to be made to the Borrowers do not and will not render either Borrower insolvent or unable to pay its debts as they become due; the capital of each Borrower and each Subsidiary is not unreasonably small, compared with the capital of companies of similar size and engaged in similar activities, in light of the needs and anticipated needs for capital of the businesses conducted or anticipated to be conducted by it; neither Borrower nor any Subsidiary is contemplating either the filing of a petition by it under any state or federal bankruptcy or insolvency laws or the liquidating of all or a major portion of its property, and none of the Borrowers' officers has knowledge of any person contemplating the filing of any such petition against either Borrower or any Subsidiary or any of its properties and assets.
- 3.10 Full Disclosure. No statement of fact made by or on behalf of either Borrower or any Subsidiary in any Loan Document or other document provided to Bank contains any untrue statement of a material fact or omits to state any material fact necessary to make statements contained therein or herein not misleading. There is no fact presently known to any of the Borrowers' partners or officers which has not been disclosed to the Bank which materially affects adversely or, as far as such officer or partner can foresee, will materially affect adversely the assets, business or condition (financial or otherwise) of either Borrower or any Subsidiary. As of the date of this Agreement, no Default exists.
- 3.11 Margin Stock. Neither Borrower owns or has any present intention of acquiring, and no portion of any Loan is to be used for the "purpose of purchasing or carrying", any "margin stock" as such terms are used in Regulations U and X of the Board of Governors of the Federal Reserve System, 12 C.F.R. 221 and 224, as amended.
  - 3.12 Patents, Trademarks, Etc. Each Borrower and each

Subsidiary owns or possesses all the patents, trademarks, service marks, trade names, copyrights, consents and licenses, and all rights with respect to the foregoing, necessary for the current and currently planned future conduct of its business, without any known conflict with the rights of others.

3.13 ERISA. All contributions and other payments required to be made by either Borrower or any entity with which either Borrower would be aggregated (a "Commonly Controlled Entity") under Section 414(b), (c), (m), or (o) of the Code to all employee benefit plans, as defined in Section 3(3) of ERISA which either Borrower or any Commonly Controlled Entity maintains or maintained or to which any of them contributes or has contributed (the "Employee Benefit Plans") have been made or reserves adequate for such purposes have been set aside and reflected on Borrower's financial statements. No Employee Benefit Plan which is subject to Section 412 of the Code or Section 302 of ERISA has an accumulated or waived funding deficiency, has permitted decreases in its funding standard account, or has applied for a waiver of the minimum funding standard or an extension of any amortization period. No proceedings have been instituted to terminate, or to appoint a trustee to administer, any Employee Benefit Plan subject to Section 412 of the Code of Section 302 of ERISA. No lien imposed under the Code or ERISA on the assets of the Borrower or any Commonly Controlled Entity exists, or is likely to arise, on account of any Employee Benefit Plan subject to Section 412 of the Code of Section 302 of ERISA. Neither Borrower nor any Commonly Controlled Entity has incurred, or is likely to incur, any material withdrawal liability with respect to any multiemployer plan as defined in Section 4001(a)(3) of ERISA ("Multiemployer Plan"). No Multiemployer Plan is insolvent, unable to pay its debts as they become due, or in reorganization. Each Employee Benefit Plan which is an employee pension benefit plan, as defined in Section 3(2) of ERISA, has been determined to be qualified under Section 401(a) or Section 403(a) of the Code and nothing has occurred which would cause the loss of such qualification or the imposition of any tax liability or penalty under the Code or ERISA on either Borrower. Each Employee Benefit Plan has been and is administered in accordance with its terms and in material compliance with applicable law. Neither Borrower nor any Commonly Controlled Entity has breached any fiduciary duty imposed on it under ERISA with respect to any Employee Benefit Plan or has engaged in any prohibited transaction, as defined in Title I of ERISA or Section 4975 of the Code, involving any Employee Benefit Plan for which no exemption is available. No Employee Benefit Plan which is an employee welfare benefit plan, as defined in Section 3(1) of ERISA, provides for continuing benefits or coverage for any participant (or beneficiary) after the termination of the participant's employment except as may be required under Section 4980B of the Code or applicable state statutory law.

#### 3.14 Environmental Laws.

(a) Each Borrower and each Subsidiary has obtained all permits, licenses and other authorizations which are required under all Environmental Laws, except to the extent failure to have any such permit, license or authorization would not have a material adverse effect on either Borrower's or any Subsidiary's business, financial condition or operations. Each Borrower and each Subsidiary is in compliance with the terms and conditions of all such permits, licenses and authorizations, and is also in compliance with all other limitations, restrictions, conditions, standards, prohibitions, requirements, obligations, schedules and timetables contained in any applicable Environmental Law or in any regulation, code, plan, order, decree, judgment, injunction, notice or demand letter issued, entered, promulgated or approved thereunder, except to the extent failure to comply would not have

a material adverse effect on either Borrower's or any Subsidiary's business, financial condition, prospects or operations.

- (b) No notice, notification, demand, request for information, citation, summons or order has been issued, no complaint has been filed, no penalty has been assessed and no investigation or review is pending or threatened by any governmental or other entity with respect to any alleged failure by either Borrower or any Subsidiary to have any permit, license or authorization required in connection with the conduct of its business or with respect to any Environmental Laws, including, without limitation, Environmental Laws relating to the generation, treatment, storage, recycling, transportation, disposal or release of any Hazardous Materials, except to the extent that such notice, complaint, penalty or investigation did not or could not result in the remediation of any property owned or used by either Borrower or any Subsidiary costing in excess of \$10,000 in the aggregate.
- (c) To the best of each Borrower's knowledge, no material oral or written notification of a release of a Hazardous Material has been filed by or on behalf of either Borrower or any Subsidiary and no property now or previously owned, leased or used by either Borrower or any Subsidiary is listed or proposed for listing on the National Priorities List under the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended, or on any similar state list of sites requiring investigation or clean-up.
- (d) There are no Encumbrances arising under or pursuant to any Environmental Laws or any of the real property or properties owned, leased or used by either Borrower or any Subsidiary and no governmental actions have been taken or are in process which could subject any of such properties to such liens or encumbrances or, as a result of which either Borrower or any Subsidiary would be required to place any notice or restriction relating to the presence of Hazardous Materials at any property owned by it in any deed to such property.
- (e) Except as disclosed on Exhibit 3.14, neither Borrower nor any Subsidiary, to the best knowledge of either Borrower, any previous owner, tenant, occupant or user of any property owned, leased or used by either Borrower or any Subsidiary has (i) engaged in or permitted any operations or activities upon or any use or occupancy of such property, or any portion thereof, for the purpose of or in any way involving the handling, manufacture, treatment, storage, use, generation, release, discharge, refining, dumping or disposal (whether legal or illegal, accidental or intentional) of any Hazardous Materials on, under, in or about such property, except to the extent commonly used in day-to-day operations of such property and in such case only in compliance with all Environmental Laws, or (ii) transported any Hazardous Materials to, from or across such property except to the extent commonly used in day-to-day operations of such property and, in such case, in compliance with, all Environmental Laws; nor, to the best knowledge of either Borrower, have any Hazardous Materials migrated from other properties upon, about or beneath such property, nor, to the best knowledge of either Borrower, are any Hazardous Materials presently constructed, deposited, stored or otherwise located on, under, in or about such property except to the extent commonly used in day-to-day operations of such property and, in such case, in compliance with, all Environmental Laws.
- 3.15 Ownership. The Corporation is the owner, directly or thorough wholly-owned Subsidiaries, beneficially and legally, of all of the outstanding equity interests in the Partnership.

Exhibit 3.15 sets forth in detail such direct and indirect ownership. No person or entity not identified on Exhibit 3.15 has an equity interest in the Partnership or any right to acquire such an interest. The Corporation's authorized capital stock consists of two classes, a limited voting rights Class A Common Stock and a full voting rights Class B Common Stock. Among its voting rights, the holders of the Class A Common Stock have the right to elect two of the Corporation's seven Directors, with the remaining five Directors elected by the holders of the Class B Common Stock. C. James Koch is currently the sole holder, beneficially and legally, of all of the issued and outstanding shares of the Class B Common Stock. The Class A Common Stock is publicly traded and is listed on the New York Stock Exchange.

#### 4. AFFIRMATIVE COVENANTS.

Each of the Borrowers covenants and agrees that, so long as the Bank has any commitment to lend hereunder or in connection with any other Obligation, or any Loan or other Obligation remains outstanding, each of the Borrowers will, and will cause each Subsidiary to:

- 4.1 Financial Reporting. Furnish to the Bank:
- (a) Within 120 days after the end of each fiscal year, its consolidated balance sheet as of the end of, and the related statements of operations and cash flow for, such year, and comparisons to prior years, audited and certified by Coopers & Lybrand (or other independent certified public accountants acceptable to the Bank), together with related consolidating statements, and promptly (and in any event within five days) after the same is available, a copy of the management letter prepared by such accountants with respect thereto;
- (b) Within 55 days after the end of each fiscal quarter, its consolidated balance sheet as of the end of, and the related statements of operations and cash flow for, the period then ended and year-to-date, and comparisons to prior periods and years, certified by the Borrowers' chief financial officer (subject to normal, recurring year-end adjustments that shall not in the aggregate be material in amount);
- (c) On the date hereof and concurrently with the delivery of the annual and quarterly financial statements required by Sections 4.1(a) and (b), a report substantially in the form of Exhibit 4.1 hereto signed on behalf of the Borrower by its chief financial officer;
- (d) No later than 60 days after the beginning of each fiscal year, financial projections of the Borrowers for such fiscal year;
- (e) Immediately upon becoming aware of the existence of any condition or event that constitutes a Default, written notice thereof specifying the nature and duration thereof and the action being or proposed to be taken with respect thereto;
- (f) Promptly (and in any event within five days) after becoming aware of any litigation or of any investigative proceedings by a governmental agency or authority commenced or threatened against either Borrower or any Subsidiary , the outcome of which would have a reasonable likelihood of resulting in a materially adverse effect on any of its material assets or its business or financial condition, written notice thereof and of the action being or proposed to be taken with respect thereto;
- (g) Promptly after the same are available, copies of all financial statements, proxy statements and regular and periodic

reports as the Partnership shall send to its partners or as the Corporation may send to its stockholders or file with the SEC or any other governmental authority at any time having jurisdiction over either Borrower;

- (h) If and when either Borrower gives or is required to give notice to the PBGC of any "Reportable Event" (as defined in Section 4043 of ERISA) with respect to any Employee Benefit Plan that might constitute grounds for a termination of such Employee Benefit Plan under Title IV of ERISA, or knows that any member of the Commonly Controlled Entity or the Employee Benefit Plan Administrator of any Employee Benefit Plan has given or is required to give notice of any such Reportable Event, a copy of the notice of such Reportable Event given or required to be given to the PBGC;
- (i) Promptly upon becoming aware of any investigative proceedings by a governmental agency or authority commenced or threatened against either Borrower or any Subsidiary regarding any potential violation of Environmental Laws or any spill, release, discharge or disposal of any Hazardous Material, written notice thereof and the action being or proposed to be taken with respect thereto; and
- (j) From time to time, such other data and information (financial and otherwise) about either Borrower or any Subsidiary and its assets and properties as the Bank may reasonably request.
- 4.2 Conduct of Business. (a) As to the Partnership, maintain in good standing its existence as a Massachusetts limited partnership and its foreign qualifications as required by applicable law where the failure to do so qualify could have a material adverse effect on it and, as to the Corporation, maintain in good standing its existence as a Massachusetts corporation and its foreign qualifications as required by applicable law where the failure to so qualify could have a material adverse effect on it, (b) as to any Subsidiary, maintain in good standing its existence as a corporation under its state of incorporation and its foreign qualifications as required by applicable law where the failure to so qualify could have a material adverse effect on it, (c) duly observe and comply in all material respects with all applicable laws and all applicable requirements of any governmental authorities whether now in effect or hereafter enacted or promulgated, including without limitation Environmental Laws, SEC laws, rules and regulations and ERISA, (d) comply in all material respects with all material agreements to which it is or becomes a party, (e) maintain and keep in full force and effect all patents, trademarks, licenses, permits and other intangibles necessary in any material respect to the proper conduct of its business, and (f) remain and engage in substantially the same business as that in which each is now engaged.
- 4.3 Maintenance and Insurance. Maintain its properties in good repair, working order and condition as required for the normal conduct of its business; and at all times maintain liability, product liability, builder's risk, workers' compensation, casualty and other insurance insuring such Borrower or Subsidiary and its properties with financially sound and reputable insurers in such amounts as the officers or partners of such Borrower or Subsidiary in the exercise of their reasonable judgment deem to be adequate and, in any event, in customary amounts with respect to like properties or risks of companies conducting similar businesses; at the Bank's request from time to time, deliver certificates of insurance evidencing such coverage.
- $4.4\,$  Taxes and Accounts Payable. Pay or cause to be paid all taxes, assessments or governmental charges on it or its

respective properties, as well as all lawful claims for labor, materials and supplies or otherwise, which, if unpaid, could become a lien or charge upon such properties or any part thereof when or before they become due; provided that this covenant shall not apply to any tax, assessment or charge that is being contested in good faith by appropriate proceedings and with respect to which adequate reserves have been established and are being maintained in accordance with generally accepted accounting principles.

- 4.5 Maintenance of Books and Records. Keep adequate books and records of account, in which true and complete entries will be made reflecting all of its business and financial transactions and prepare financial statements hereunder in accordance with U.S. generally accepted accounting principles consistently applied and applicable law. Permit agents or representatives of the Bank, at any reasonable time, and upon reasonable notice (or if a Default shall have occurred and is continuing, at any time without prior notice), to (a) examine, make copies of and take abstracts from the financial and accounting books and records of the Borrowers, and (b) discuss the affairs, finances and accounts of the Borrowers with their officers, partners, employees and accountants.
- 4.6 Environmental Laws. Comply in all material respects with, and perform all Clean-up Work necessary under, all Environmental Laws applicable (now or in the future) to either Borrower or any Subsidiary or its business.
- 4.7 Further Assurances. At any time and from time to time, (a) cause any Subsidiary prior to or upon becoming a Material Subsidiary to execute and deliver to Bank an unlimited guaranty in favor of Bank, guaranteeing all existing and future obligations of Borrowers to Bank; and (b) execute and deliver such further instruments and take such further action as may reasonably be requested by the Bank to effect the purposes of the Loan Documents.
- 4.8 Principal Depository. Use the Bank as its principal depository.

# 5. FINANCIAL COVENANTS.

- 5.1 Covenants. Each of the Borrowers covenants and agrees that, so long as the Bank has any commitment to lend hereunder or any Loan or other Obligation to the Bank remains outstanding, each of the Borrowers will:
- (a) Maintain at all times during the period from December 31, 1996 through March 31, 1997 a Tangible Net Worth of the Corporation of not less than \$50,000,000; thereafter maintain Tangible Net Worth of the Corporation at all times during each fiscal quarter of at least (i) the minimum amount of Tangible Net Worth required hereunder as of the last day of the immediately preceding fiscal quarter, plus (ii) 50% of the positive Net Income (with no reduction for losses) for such immediately preceding fiscal quarter; and
- (b) Not permit at any time the ratio of Total Liabilities of the Corporation to Tangible Net Worth of the Corporation to be greater than 1.00:1.00;
- (c) Not permit the ratio of Total Funded Debt of the Corporation at any time to EBITDA of the Corporation for the most recently ended period of four consecutive fiscal quarters to be greater than 2.50:1.00; and
  - (d) As to the Corporation, earn Net Income of at least

- 5.2 Definitions. For purposes of this Section, the following terms shall have the following meanings and any undefined terms shall be defined in accordance with generally accepted accounting principles on a basis consistent with (a) that used in prior periods and (b) that in preparing the audited financial statements referred to in Sections 3.5 and 4.1(a):
- (a) As used herein, "EBITDA" shall mean, for any period, an amount equal to Net Income of the Corporation for such period plus, to the extent deducted in the calculation of Net Income for such period (i) taxes in respect of income and profits paid or accrued by the Corporation during such period (ii) Interest Expense for such period and (iii) depreciation, amortization and other non-cash charges taken in accordance with generally accepted accounting principles for such period.
- (b) As used herein, "Indebtedness for Borrowed Money" shall mean, without duplication, all indebtedness with respect to any of the following of the Corporation: (i) money borrowed (whether recourse or nonrecourse), including principal, interest and premiums, (ii) obligations evidenced by a bond, debenture, note or other similar written obligation to pay money, (iii) capitalized lease obligations, (iv) obligations under conditional sales or other title retention agreements or secured by any Encumbrance, (v) any letters of credit or similar instruments (including reimbursement obligations with respect thereto), (vi) the deferred purchase price of property or services, except trade payables and accrued expenses as incurred in the ordinary course of business, or (vii) any guaranty of any or all of the foregoing, in case each determined in accordance with generally accepted accounting principles.
- (c) As used herein, "Interest Expense" shall mean, for any period, interest expense (including imputed interest on capital lease obligations) of the Corporation for such period.
- (d) As used herein, "Net Income" shall mean the net income (or loss) of the Corporation, after deducting all operating expenses, provisions for all taxes and reserves (including reserves for deferred income taxes) and all other proper deductions, all determined in accordance with generally accepted accounting principles.
- (e) As used herein, "Tangible Net Worth" shall mean as of any date as of which the amount thereof shall be determined, the total assets of the Corporation minus (a) the sum of any amounts attributable to (i) goodwill, (ii) intangible items such as unamortized debt discount and expense, patents, trade and service marks and names, copyrights and research and development expenses except prepaid expenses, (iii) all reserves not already deducted from assets, (iv) any write up in the book value of assets resulting from any revaluation thereof subsequent to the date of the Corporation's audited financial statements referred to in Section 3.5, and (v) the value of any minority interest in any Subsidiaries (other than Permitted Acquisitions), and (b) Total Liabilities of the Corporation.
- (f) As used herein, "Total Funded Debt" shall mean at any time, all Indebtedness for Borrowed Money of the Corporation at such time, determined in accordance with generally accepted accounting principles. The aggregate amount of Total Funded Debt at any time shall include all accrued interest which has become due and payable but has not been paid (whether or not capitalized).
  - (g) As used herein, "Total Liabilities" shall mean at any

date as of which the amount thereof shall be determined, all obligations which should, in accordance with generally accepted accounting principles, be classified as liabilities on the balance sheet of the Corporation, including in any event all Indebtedness.

#### NEGATIVE COVENANTS.

Each of the Borrowers covenants and agrees that, so long as the Bank has any commitment to lend hereunder or any Loan or other Obligation to the Bank remains outstanding, each of the Borrowers will not, and will cause each of the Subsidiaries to not:

- 6.1 Indebtedness, Guarantees. Create, incur, assume or be or remain liable with respect to any Indebtedness or Guarantees, other than the following:
- (a) Indebtedness of the Borrowers to the Bank or any of its affiliates;
- (b) Indebtedness existing as of the date of this Agreement in an outstanding principal amount of not greater than \$2,200,000 under that certain Loan, Security and Trust Agreement, dated as of October 1, 1987 among the Partnership, The First National Bank of Boston, as trustee, and the Massachusetts Industrial Finance Agency, provided such Indebtedness shall not be modified or amended, nor shall payment thereof be extended without the prior written consent of the Bank;
- (c) Purchase money Indebtedness permitted under Section  $6.2\,\mathrm{(c)}$ ;
- (d) Guarantees in favor of the Bank or any of its affiliates; and
- (e) Guarantees resulting from the endorsement of negotiable instruments for collection in the ordinary course of business.
- 6.2 Encumbrances. Create, incur, assume or suffer to exist any mortgage, pledge, security interest, lien or other charge or encumbrance, including the lien or retained security title of a conditional vendor upon or with respect to any of its property or assets ("Encumbrances"), or assign or otherwise convey any right to receive income, except the following ("Permitted Encumbrances"):
- (a) landlords' and lessors' liens in respect of rent not in default or liens in respect of pledges or deposits under workers' compensation, unemployment insurance, social security laws, or similar legislation (other than ERISA) or in connection with appeal and similar bonds incidental to litigation; mechanics' and similar liens, if the obligations secured by such liens are not then delinquent; liens securing the performance of bids, tenders, contracts (other than for the payment of money); and statutory obligations incidental to the conduct of its business that do not in the aggregate materially detract from the value of the property of the Borrower, or materially impair the use thereof in the operation of the business of the Borrower;
- (b) judgment liens that shall not have been in existence for a period longer than 30 days after the creation thereof or, if a stay of execution shall have been obtained, for a period longer than 30 days after the expiration of such stay;
- (c) Encumbrances against either Borrower or any Subsidiary securing purchase money Indebtedness for tangible personal property (other than inventory) in the ordinary course of

business in an aggregate amount (for the Borrowers and the Subsidiaries as a group) of no more than \$5,000,000, provided that (i) the obligations secured by any such Encumbrance shall not exceed 100% of the lesser of cost or fair market value as of the time of the acquisition of such property covered thereby, and (ii) each such Encumbrance shall at all times be limited solely to the item or items of property so acquired;

- (d) Encumbrances on certain assets of the Partnership and the Corporation (other than accounts and inventory and proceeds thereof) securing only the Indebtedness permitted under Section 6.1(b); and
- (e) easements, rights of way, restrictions and other similar charges or Encumbrances relating to real property and not interfering in a material way with the ordinary conduct of the business of the Borrower and the Subsidiaries.
- 6.3 Sale and Leaseback. Enter into any arrangement, directly or indirectly, whereby it shall sell or transfer any property owned by it, in order to lease such property or lease other property that either Borrower or any Subsidiary intends to use for substantially the same purpose as the property being sold or transferred.
- 6.4 Merger; Consolidation; Disposition of Assets; Etc. Without the Bank's prior written consent: sell, lease or otherwise dispose of any of its assets or properties except inventory and/or equipment in the ordinary course of business; amend or permit the amendment of its partnership agreement or certificate of limited partnership or charter or bylaws in any way that could have a material adverse effect upon its business or the Bank's rights under any Obligation or alter the relative voting rights of the Class A and Class B Common Stock of the Corporation (whether for the Board of Directors or otherwise); dissolve, liquidate, merge or consolidate into or with any other person or entity; form (by spin-off or otherwise) any Subsidiary unless no Default then exists or would result therefrom and such Subsidiary , if a Material Subsidiary, executes and delivers in favor of the Bank an unlimited guaranty, in form satisfactory to the Bank, guaranteeing all existing and future obligations of the Borrowers to the Bank; or change its fiscal year.
- 6.5 Investments. Make or maintain any Investments other than Permitted Investments.
- 6.6 Transactions with Affiliates. Enter into any transaction, including, without limitation, the purchase, sale or exchange of property or assets or the rendering or accepting of any service with or to any Affiliate of either Borrower or any Subsidiary except in the ordinary course of business and pursuant to the reasonable requirements of such Borrower's business or Subsidiary's and upon terms not less favorable to the Borrowers and the Material Subsidiaries than they could obtain in a comparable arm's-length transaction with a third party other than such Affiliate.
- 6.7 Distributions. Without the Bank's prior written consent, no Borrower shall make any distributions on account of or in redemption, retirement or purchase its capital stock, partnership or other equity interests; except that (a) the Partnership may make distributions on account of its partnership interests to the Corporation, (b) the Subsidiaries may make distributions on account of its capital stock to the Borrowers, and (c) the Corporation may redeem from former employees nonvested shares in the Corporation's Class A Common Stock issued under the Corporation's Employee Equity Incentive Plan, to the extent that the Corporation has the right to redeem such shares

under said Plan.

#### 7. DEFAULTS.

- 7.1 Events of Default. There shall be an Event of Default hereunder if any of the following events occurs:
- (a) default in the payment or mandatory prepayment of any amount of principal, interest, fees or expenses of any Loan or any other Obligation when due; or
- (b) failure to perform any term, covenant or agreement contained in Section 4, Section 5, Section 6 or Section 8.2; or
- (c) failure by either Borrower or any Subsidiary, as applicable, to perform any other term, covenant or agreement contained in any Loan Document if such default shall continue for 30 days after notice thereof is sent to the Borrowers by the Bank; or
- (d) any representation or warranty made in this Agreement or any certificate or other document furnished by or on behalf of either Borrower or any Subsidiary at the closing hereunder shall prove to have been false in any material respect upon the date when made or deemed to have been made; or
- (e) default or breach (after the applicable grace period, if any) under the Indebtedness described in Section 6.1(b) or under any instrument or agreement of either Borrower or any Subsidiary with respect to other Indebtedness of either Borrower or any Subsidiary (other than to the Bank) for borrowed money or with respect to financing available to either Borrower or any Subsidiary (other than from the Bank), if the effect of such default or breach is to permit the holder of such Indebtedness to accelerate the maturity thereof; or
- (f) either Borrower or any Subsidiary shall (i) apply for or consent to the appointment of, or the taking of possession by, a receiver, custodian, trustee, liquidator or similar official of itself or of all or a substantial part of its property, (ii) be generally not paying its debts as such debts become due, (iii) make a general assignment for the benefit of its creditors, (iv) commence a voluntary case under the Federal Bankruptcy Code (as now or hereafter in effect), (v) take any action or commence any case or proceeding under any law relating to bankruptcy, insolvency, reorganization, winding-up or composition or adjustment of debts, or any other law providing for the relief of debtors, (vi) fail to contest in a timely or appropriate manner, or acquiesce in writing to, any petition filed against it in an involuntary case under the Federal Bankruptcy Code or other law, (vii) take any action under the laws of its jurisdiction of incorporation or organization similar to any of the foregoing, or (viii) take any corporate action for the purpose of effecting any of the foregoing; or
- (g) a proceeding or case shall be commenced, without the application or consent of either Borrower or any Subsidiary in any court of competent jurisdiction, seeking (i) the liquidation, reorganization, dissolution, winding up, or composition or readjustment of such Borrower's or such Subsidiary's debts, (ii) the appointment of a trustee, receiver, custodian, liquidator or the like of all or any substantial part of either Borrower's or any Subsidiary's assets, or (iii) similar relief in respect of either Borrower or any Subsidiary, under any law relating to bankruptcy, insolvency, reorganization, winding-up or composition or adjustment of debts or any other law providing for the relief of debtors, and such proceeding or case shall continue undismissed, or unstayed and in effect, for a period of 60 days;

or an order for relief shall be entered in an involuntary case under the Federal Bankruptcy Code, against either Borrower; or action under the laws of the jurisdiction of incorporation or organization similar to any of the foregoing shall be taken with respect to either Borrower or any Subsidiary and shall continue unstayed and in effect for any period of 60 days; or

- (h) a judgment or order for the payment of money shall be entered against either Borrower or any Subsidiary by any court, or a warrant of attachment or execution or similar process shall be issued or levied against property of either Borrower or any Subsidiary, that in the aggregate exceeds \$500,000 in value and such judgment, order, warrant or process shall continue undischarged or unstayed for 60 days or execution shall have issued in respect thereto; or
- (i) there shall occur any material adverse change in the assets, liabilities, financial condition, business or prospects of either Borrower or any Subsidiary, as reasonably determined by the Bank acting in good faith; or
- (j) for any reason, any Loan Document at any time shall not be in full force and effect in all material respects or shall not be enforceable in all material respects in accordance with its terms, or any Encumbrance granted pursuant thereto shall fail to be perfected, or any person or entity (other than the Bank) shall contest the validity, enforceability or perfection of any Encumbrance granted pursuant thereto, or any party thereto (other than the Bank) shall seek to disaffirm, terminate, limit or reduce its obligations under any of the Loan Documents; or
- (k) for any reason, C. James Koch shall cease to be actively engaged on a full-time basis in the management of the Borrowers, unless a successor or successors with qualifications, experience and reputation within the consumer beverage business reasonably acceptable to the Bank is appointed and commences management of the Borrowers within 120 days after such cessation; or
- (1) for any reason, the Partnership's existing general partner shall cease to be the sole general partner of the Partnership; or
- (m) for any reason, the Corporation shall cease to be the beneficial and legal owner of all classes of voting and nonvoting equity interests of the Partnership; or C. James Koch shall cease to be the beneficial and legal owner of at least 51% of all issued and outstanding Class B Common Stock of the Corporation on a fully diluted basis; or no Class B Common Stock of the Corporation shall be issued and outstanding.
- 7.2 Remedies. Upon the occurrence of an Event of Default described in Sections 7.1(f) and (g), immediately and automatically, and upon the occurrence of any other Event of Default, at any time thereafter while such Event of Default is continuing, at the Bank's option and upon the Bank's declaration, (a) the Commitment and all other commitments of the Bank to lend to the Borrowers shall terminate; (b) the unpaid principal amount of the Loans together with accrued interest and all other Obligations shall become immediately due and payable without presentment, demand, protest or further notice of any kind, all of which are hereby expressly waived; and (c) the Bank may exercise any and all rights it has under the Loan Documents or at law or in equity, and proceed to protect and enforce the Bank's rights by any action at law, in equity or other appropriate proceeding.

8.1 Notices. Unless otherwise specified herein, all notices hereunder to any party hereto shall be in writing, shall be addressed to such party at its address indicated below, and shall be deemed to have been given when delivered by hand, two days after being properly deposited in the mails postage prepaid, when sent by electronic telecopy transmission, or one day after delivery to an overnight courier:

## If to the Borrowers:

The Boston Beer Company, Inc.
Boston Beer Company Limited Partnership
75 Arlington Street
Boston, MA 02116
Attn: Alfred W. Rossow, Chief Operating Officer
Telecopy No.: (617) 482-9210

with a copy to:

Frederick H. Grein, Jr., Esq. Hutchins, Wheeler & Dittmar 101 Federal Street Boston, MA 02110 Telecopy No.: (617) 951-1295

## If to the Bank:

Fleet National Bank
One Federal Street
Boston, MA 02110
Attn: Mary Barcus, VP
Telecopy No.: (617) 346-5788

with a copy to:

Susan E. Siebert, Esq. Edwards & Angell 101 Federal Street Boston, MA 02110 Telecopy No.: (617) 439-4170

or at any other address specified by such party in writing.

- $8.2\,$  Expenses. Pay to the Bank on demand all Expenses incurred by the Bank.
- 8.3 Set-Off. Regardless of the adequacy of any collateral (whether or not fully secured), if any, or other means of obtaining repayment of the Obligations hereunder and any other Obligations to the Bank, any deposits, balances or other sums credited by or due from the head office of the Bank or any of its branch offices to either Borrower or any Subsidiary may, at any time and from time to time after the occurrence of an Event of Default hereunder, without notice to the Borrowers or compliance with any other condition precedent now or hereafter imposed by statute, rule of law, or otherwise (all of which are hereby expressly waived) be set off, appropriated, and applied by the Bank against any and all Obligations of either Borrower or any Subsidiary to the Bank, regardless of whether such Obligations have matured, in such manner as the Bank in its sole discretion may determine.
- 8.4 No Waivers; Term. No failure or delay by the Bank in exercising any right, power or privilege hereunder or under any other Loan Document shall operate as a waiver thereof; nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right,

power or privilege. The rights and remedies herein and in the other Loan Documents are cumulative and not exclusive of any rights or remedies otherwise provided by agreement or law. This Agreement shall continue in force and effect so long as the Bank has any commitment to make Loans hereunder or any Loan or any other Obligation shall be outstanding.

- 8.5 Governing Law. This Agreement, the Notes and the other Loan Documents shall be deemed to be contracts made under seal and shall be construed in accordance with and governed by the laws of the Commonwealth of Massachusetts (without giving effect to any conflicts of laws provisions contained therein).
- 8.6 Amendments. Neither this Agreement nor the Notes nor any other Loan Document nor any provision hereof or thereof may be amended or waived except by a written instrument signed by the Bank and, in the case of amendments, by the Bank and the other parties named thereto.
- 8.7 Binding Effect of Agreement. This Agreement shall be binding upon and inure to the benefit of the Borrowers and the Bank and their respective successors and assigns; provided that neither Borrower may assign or transfer its rights or obligations hereunder.
- 8.8 Entire Agreement. The Loan Documents constitute the final agreement of the parties hereto and supersede any prior agreement or understanding, written or oral, with respect to the matters contained herein and therein.
- 8.9 Counterparts, Etc. This Agreement may be signed in any number of counterparts with the same effect as if the signatures hereto and thereto were upon the same instrument. The invalidity or unenforceability of any one or more phrases, clauses or sections of this Agreement shall not affect the validity or enforceability of the remaining portions hereof. The captions and headings of the various sections and subsections of this Agreement are provided for convenience only and shall not be construed to modify the meaning of such sections or subsections.
- 8.10 WAIVER OF JURY TRIAL. EACH OF THE BORROWERS AND THE BANK AGREES THAT IT SHALL NOT (A) SEEK A JURY TRIAL IN ANY LAWSUIT, PROCEEDING, COUNTERCLAIM OR ANY OTHER ACTION BASED UPON, OR ARISING OUT OF, THIS AGREEMENT, OR ANY OTHER LOAN DOCUMENT OR OBLIGATION, OR (B) SEEK TO CONSOLIDATE ANY SUCH ACTION WITH ANY OTHER ACTION IN WHICH A JURY TRIAL CANNOT BE OR HAS NOT BEEN WAIVED. THE PROVISIONS OF THIS PARAGRAPH HAVE BEEN FULLY DISCUSSED BY THE BANK AND THE BORROWERS AND SHALL BE SUBJECT TO NO EXCEPTIONS. THE BANK HAS NOT AGREED WITH OR REPRESENTED TO THE BORROWERS THAT THE PROVISIONS OF THIS SECTION WILL NOT BE FULLY ENFORCED IN ALL INSTANCES.
- 8.11 CONSENT TO JURISDICTION. EACH OF THE BORROWERS HEREBY SUBMITS TO THE JURISDICTION OF THE COURTS OF THE COMMONWEALTH OF MASSACHUSETTS AND THE UNITED STATES DISTRICT COURT FOR THE DISTRICT OF MASSACHUSETTS, AS WELL AS TO THE JURISDICTION OF ALL COURTS FROM WHICH AN APPEAL MAY BE TAKEN OR OTHER REVIEW SOUGHT FROM THE AFORESAID COURTS, FOR THE PURPOSE OF ANY SUIT, ACTION OR OTHER PROCEEDING ARISING OUT OF EITHER BORROWER'S OBLIGATIONS UNDER OR WITH RESPECT TO ANY LOAN DOCUMENT OR ANY OF THE TRANSACTIONS CONTEMPLATED HEREBY OR ANY OTHER OBLIGATION TO BANK, AND EXPRESSLY WAIVES ANY AND ALL OBJECTIONS IT MAY HAVE AS TO VENUE IN ANY OF SUCH COURTS.
- 8.12 Indemnification; Joint and Several Liability. Each of the Borrowers will indemnify and hold the Bank harmless against any and all claims, demands, suits, actions, causes of action, damages, losses, settlement payments, obligations, costs,

expenses and all other liabilities whatsoever which shall at any time or times be incurred or sustained by the Bank or by any of its shareholders, directors, officers, employees, subsidiaries, affiliates or agents (except any of the foregoing incurred or sustained as a result of the gross negligence or willful misconduct of the Bank) on account of, or in relation to, or in any way in connection with, any of the arrangements or transactions contemplated by, associated with or ancillary to any of the Obligations. All of the obligations of the Borrowers under any and all of the Loan Documents are and shall be joint and several.

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed under seal by their duly authorized officers as of the day and year first above written.

THE BOSTON BEER COMPANY, INC.

By: Title:

BOSTON BEER COMPANY LIMITED PARTNERSHIP

By The Boston Brewing Company,
Inc.,
 Its General Partner

By: Title:

FLEET NATIONAL BANK

By:
Mary Barcus, Vice President

Schedule A

## General Defined Terms

As used herein, the following terms shall have the following meanings and any undefined accounting terms shall be defined in accordance with generally accepted accounting principles on a basis consistent with (a) that used in prior periods and (b) that used in preparing the audited financial statements referred to in Sections 3.5 and 4.1(a):

Acquisition Loans: See Section 1.10.

Adjusted Libor Rate: See Schedule B.

Affiliate: as applied to the Borrower, any member, trustee, director, officer, manager or employee, any corporation, association, partnership, firm or other entity in which the Borrower is a partner, shareholder (except any holder of not more than 5% of the outstanding stock of any company listed on a national securities exchange or actively traded on the over-the-

counter securities market), warrant holder, option holder, member, director or officer, and any other person or entity directly or indirectly controlling, controlled by, or under common control with, the Borrower.

Alternative Prime Rate: The higher of (i) the Prime Rate, and (ii) the Federal Funds Rate plus 1/2 of 1% per annum (rounded upwards, if necessary, to the next 1/8 of 1%). The Alternative Prime Rate is not necessarily intended to be the lowest rate of interest determined by the Bank in connection with extensions of credit.

Bank: see Preamble.

Borrower and Borrowers: see Preamble.

Business Day: (i) for all purposes other than as covered by clause (ii) below, any day other than a Saturday, Sunday or legal holiday on which banks in Boston, Massachusetts are open for the conduct of a substantial part of their commercial banking business; and (ii) with respect to all notices and determinations in connection with, and payments of principal and interest on, Libor Loans, any day that is a Business Day described in clause (i) and also a day for trading by and between banks in U.S. Dollar deposits in the interbank London market.

Clean-up Work: All activities, including, without limitation, cleanup design and implementation, removal activities, investigation, field and laboratory testing and analysis, monitoring and other remedial and response actions, taken or to be taken, arising out of or in connection with Hazardous Materials, including without limitation (a) all activities included within the meaning of the terms "removal," "remedial action" or "response", as defined in 42 U.S.C. Section 9601(23), (24) and (25), and (b) all activities included within the meaning of the terms "remedial response actions" and "Remedial Response Implementation Plan (RRIP)," as defined in 310 CMR 40.

Code: the Internal Revenue Code of 1986 and the rules and regulations thereunder, as may be supplemented or amended from time to time.

Commitments: The Revolver A Commitment and the Revolver B Commitment.

Commonly Controlled Entity: See Section 3.13.

Conversion Date: See Section 1.1(a).

Default: an Event of Default or event or condition that, but for the requirement that time elapse or notice be given, or both, would constitute an Event of Default.

EBITDA: See Section 5.2(c).

Employee Benefit Plan: See Section 3.13.

Encumbrance: See Section 6.2.

ERISA: the Employee Retirement Income Security Act of 1974 and the rules and regulations thereunder, as supplemented or amended from time to time.

Environmental Laws: any and all applicable foreign, federal, state and local environmental, health or safety statutes, laws, regulations, rules, ordinances, policies and rules or common law (whether now existing or hereafter enacted or

promulgated), of all governmental agencies, bureaus or departments which may now or hereafter have jurisdiction over the Borrower and all applicable judicial and administrative and regulatory decrees, judgments and orders, including common law rulings and determinations, relating to injury to, or the protection of, human health or the environment, including, without limitation, all requirements pertaining to reporting, licensing, permitting, investigation, remediation and removal of emissions, discharges, releases or threatened releases of Hazardous Materials, chemical substances, pollutants or contaminants whether solid, liquid or gaseous in nature, into the environment or relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of such Hazardous Materials, chemical substances, pollutants or contaminants.

Expenses: any and all present and future expenses incurred by or on behalf of the Bank in connection with this Agreement or any of the other Loan Documents, whether incurred heretofore or hereafter, which expenses shall include, without being limited to, the cost of record searches, attorneys' fees (for regulation, preparation, execution and delivery of the Loan Documents and amendments thereto and enforcement and collection and bankruptcy services with respect thereto), disbursements and expenses, all costs and expenses incurred by Bank in opening bank accounts, depositing checks, receiving and transferring funds, and any charges due to "insufficient funds" of deposited checks and the Bank's standard fee relating thereto, the fees payable to the collateral examination staff of the Bank, fees and expenses of accountants, appraisers or other experts or advisors retained by the Bank, fees and taxes relative to the filing of financing statements, costs of preparing and recording the Loan Documents and all other expenses, costs and fees contemplated by this Agreement.

Event of Default: any event described in Section 7.1.

Expiration Date: See Section 1.1(a).

Federal Funds Rate: For any day, a fluctuating interest rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers, as published for such day (or, if such day is not a Business Day, for the next preceding Business Day) by the Federal Reserve Bank of New York, or, if such rate is not so published for any day that is a Business Day, the average of the quotations for such day on such transactions received by the Bank from three Federal funds brokers of recognized standing selected by the Bank.

Hazardous Material: any material or substance (a) which, whether by its nature or use, is now or hereafter defined as a hazardous waste, hazardous substance, hazardous material, pollutant or contaminant under any Environmental Law, including, but not limited to, the Comprehensive Environmental Response, Compensation and Liability Act of 1980 ("CERCLA"), 42 U.S.C. 9601(14) and (33), the Resource Conservation and Recovery Act ("RCRA"), 42 U.S.C. 6903(5), and the Massachusetts Oil and Hazardous Material Chapter 21E, 2, (b) which is toxic, explosive, corrosive, flammable, infectious, radioactive, carcinogenic, mutagenic or otherwise hazardous to human health or the environment, (c) which is or contains petroleum or any fraction thereof, including crude oil, heating oil, gasoline or diesel fuel, or (d) the presence of which requires investigation or remediation under any Environmental Law.

Indebtedness: (a) All obligations for borrowed money or other extensions of credit whether or not secured or unsecured,

absolute or contingent, including, without limitation, unmatured reimbursement obligations with respect to letters of credit or guarantees issued for the account of or on behalf of the Borrower and all obligations representing the deferred purchase price of property, other than accounts payable arising in the ordinary course of business, (b) all obligations evidenced by bonds, notes, debentures or other similar instruments, (c) all obligations secured by any mortgage, pledge, security interest or other lien on property owned or acquired by the Borrower whether or not the obligations secured thereby shall have been assumed, and (d) all obligations arising under capital leases.

Interest Expense: See Section 5.2(c).

Investment: as applied to the Borrower, the purchase or acquisition of any share of capital stock, partnership interest, evidence of indebtedness or other equity security or all or substantially all of the assets of any other person or entity, any loan, advance or extension of credit to, or contribution to the capital of, any other person or entity, any real estate held for sale or investment, any commodities futures contracts held other than in connection with bona fide hedging transactions, any other investment in any other person or entity, and the making of any commitment or acquisition of any option to make an Investment.

Libor Base Rate: See Schedule B.

Libor Loan: any Loan bearing interest at a rate determined with reference to the Adjusted Libor Rate.

Libor Period: See Schedule B.

Libor Terms: see Section 1.3 and Schedule B.

Loan Documents: see Section 3.2 of Schedule C.

Loans: The Revolving Loans A, the Revolving Loans B and the  $\operatorname{\mathsf{Term}}$  Loan.

Material Subsidiary: At any date of determination, any Subsidiary (a) whose total assets at the last day of the most recently ended fiscal quarter have a market value equal to or greater than 5% of the consolidated total assets of Borrowers and their Subsidiaries at such date, determined in accordance with generally accepted accounting principles, (b) whose Tangible Net Worth at the last day of the most recently ended fiscal quarter is equal to or greater than 5% of the Tangible Net Worth at such date, or (c) whose gross revenues with respect to the most recently ended four-quarter period are equal to or greater than 5% of the consolidated gross revenues of the Borrowers and their Subsidiaries for such four-quarter period.

Maturity Date: See Section 1.1(c).

Multiemployer Plan: See Section 3.13.

Net Income: See Section 5.2(c).

Notes: see Section 1.2.

Notice of Borrowing or Conversion: see Section 1.4.

Obligations: any and all existing and future obligations now or hereafter owed by the Borrower to the Bank, whether direct or indirect, contingent or liquidated.

Original Agreement: See Recitals.

Original Revolving Note: See Recitals.

Partnership: See Preamble.

PBGC: the Pension Benefit Guaranty Corporation or any entity succeeding to any or all of its functions under ERISA.

Permitted Acquisitions: any acquisition of stock or assets by the Corporation which has met the following conditions: (a) the aggregate amount of proceeds of the Loans used in all such acquisitions since the date of this Agreement shall not exceed \$5,000,000, (b) the aggregate amount paid in cash or cash equivalents by the Borrowers and their Subsidiaries in connection with all such acquisitions since the date of this Agreement (whether or not proceeds of the Loans) shall not exceed \$7,500,000, (c) after giving effect to such acquisition, on a pro forma basis as of the most recently ended fiscal quarter, the Borrowers shall be in full compliance with all of its obligations set forth in Section 5 of this Agreement, (d) if such acquisition is a stock acquisition and the acquired company or companies is not being merged into the Corporation simultaneously and is or becomes a Material Subsidiary, such acquired company or companies shall prior to or upon becoming a Material Subsidiary execute an unlimited quaranty in form satisfactory to the Bank, quaranteeing all existing and future obligation of the Borrowers to the Bank, (e) prior notice of such acquisition shall have been delivered to the Bank, describing the terms of such acquisition, including the purchase price thereof, and whether the acquired company or companies are or are intended to become Material Subsidiaries, and (e) no Default shall exist hereunder or result from such acquisition. The Bank acknowledges that the Schoenling Brewery acquisition which closed in March, 1997 (and the subsequent real property acquisition relating thereto) are Permitted Acquisitions not included in the \$5,000,000 and \$7,500,000 thresholds set forth above.

Permitted Encumbrances: see Section 6.2.

Permitted Investments: as applied to either Borrower, investments (i) in notes, bonds or other obligations of the United States of America or any agency thereof that as to principal and interest constitute direct obligations of or are quaranteed by the United States of America; (ii) in certificates of deposit or other deposit instruments or accounts of the Bank or any other banks or trust companies organized under the laws of the United States or any state thereof that have capital and surplus of at least \$100,000,000; (iii) in mutual funds investing only in commercial paper that is rated not less than P-2 or A-2 or their equivalents by Moody's Investors Service, Inc. or Standard & Poor's Corporation, respectively, or their successors; (iv) up to \$7,500,000 in the aggregate in connection with Permitted Acquisitions after the date of this Agreement, (v) the investments of the Corporation in its existing Subsidiaries, and (vi) advances to employees for the purchase of stock options not to exceed \$1,000,000 outstanding at any time in the case of any one employee and not to exceed \$2,000,000 outstanding at any time in the aggregate to all employees of the Borrowers.

Prime Rate: the annual floating rate of interest announced from time to time by the Bank at its head office as its Prime Rate  ${\sf Rate}$ 

Prime Rate Loan: any Loan bearing interest determined with reference to the Alternative Prime Rate.

Revolver A Commitment: See Section 1.1(a).

Revolver B Commitment: See Section 1.1(b).

Revolving Credit Period: the period from the date hereof through and including the Conversion Date and the Expiration Date or such earlier date or dates by which both Commitments are terminated.

Revolving Loans A: See Section 1.1(a).

Revolving Loans B: See Section 1.1(b).

 $\,$  SEC: The Securities and Exchange Commission and any successor agency thereof.

Subsidiaries: (a) Any corporation, association, limited liability company, joint stock company, business trust or other similar organization of which 50% or more of the ordinary voting power for the election of a majority of the members of the board of directors or other governing body of such entity is held or controlled by either Borrower or any Subsidiary of either Borrower; (b) any other such organization the management of which is directly or indirectly controlled by either Borrower or any Subsidiary of either Borrower through the exercise of voting power or otherwise; or (c) any joint venture, association, partnership or other entity in which either Borrower or any Subsidiary of either Borrower has at least a 50% equity interest, directly or indirectly.

Tangible Net Worth: See Section 5.2(c).

Term Loan: See Section 1.1.

Total Liabilities: See Section 5.2(c).

Schedule B

Libor Terms

## A. Definitions.

As used in this Schedule A, the following terms have the following meanings:

Libor Period: With respect to each Libor Loan, the period commencing on the date such Loan is made or converted from a Prime Rate Loan, or the last day of the immediately preceding Libor Period, as to Libor Loans being continued as such, and ending one month, two months or three months thereafter, as the Borrowers may elect in the applicable Notice of Borrowing or Conversion; provided that:

- (i) any Libor Period (other than a Libor Period determined pursuant to clause (iv) below) that would otherwise end on a day that is not a Business Day shall be extended to the next succeeding Business Day unless such Business Day falls in the next calendar month, in which case such Interest Period shall end on the immediately succeeding Business Day;
- (ii) if the Borrowers shall fail to give notice as provided in Section 1.4, the Borrowers shall be deemed to have requested a conversion of the affected Libor Loan to a Prime Rate Loan on the last day of the then current Libor Period with respect thereto;
  - (iii) any Libor Period relating to a Libor Loan that begins

on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall, subject to clause (iv) below, end on the last Business Day of a calendar month;

- (iv) any Libor Period for a Revolving Loan A or Revolving Loan B must end on or before the Expiration Date and the Conversion Date, as applicable, and any Libor Period for the Term Loan must end on or before the Maturity Date;
- (v) no Libor Period shall include a principal repayment date for the Loans unless an aggregate principal amount of Loans at least equal to the principal amount due on such principal repayment date shall be Prime Rate Loans or Libor Loans having Libor Periods ending on or before such date; and
- (vi) notwithstanding clauses (iv) and (v) above, no Libor Period shall have a duration of less than one month.

Libor Base Rate: With respect to each day during each Libor Period pertaining to any Libor Loans, the interest rate per annum (rounded upward, if necessary, to the nearest 1/16th of 1%) at which the Bank is offered deposits in U.S. Dollars at or about 10:00 A.M. (Eastern Time), two London business days prior to the beginning of such Libor Period in the London interbank market for delivery on the first day of such Libor Period, for the number of days comprised therein and in an amount comparable to the amount of the Libor Loan to be outstanding during such Libor Period.

Adjusted Libor Rate: With respect to each day during each Libor Period pertaining to a Libor Loan, a rate per annum determined for such day in accordance with the following formula (rounded upward, if necessary, to the nearest 1/16th of 1%):

## Libor Base Rate

1.00 - Libor Reserve Requirements

Libor Reserve Requirements: For any day as applied to a Libor Loan, the aggregate (without duplication) of the rates (expressed as a decimal fraction) of reserve requirements in effect on such day (including without limitation basic, supplemental, marginal and emergency reserves) under any regulations of the Board of Governors of the Federal Reserve System or other governmental authority having jurisdiction with respect thereto) prescribed for eurocurrency funding (currently referred to as "Eurocurrency Liabilities" in Regulation D of such Board) maintained by a member bank of the Federal Reserve System.

Regulatory Change: Any change after the date of this Agreement in United States federal, state or foreign law or regulation (including without limitation Regulation D), including without limitation the issuance of any final regulations or guidelines, or the adoption or making after such date of any interpretation, directive or request applying to a class of banks including the Bank or under any United States federal, state or foreign law or regulations (whether or not having the force of law and whether or not failure to comply therewith would be unlawful) by any court or governmental or monetary authority charged with the interpretation thereof.

Regulation D: Regulation D of the Board of Governors of the Federal Reserve System, as the same may be amended or supplemented from time to time.

- B. General Terms.
- 1. Libor Loans. Each Libor Loan shall bear interest on the outstanding principal amount thereof, for each Libor Period applicable thereto, at a rate as set forth in Section 1.3. Such interest shall be payable for such Libor Period on the last day thereof and when such Libor Loan is due (whether at maturity, by reason of acceleration or otherwise) and, if such Libor Period is longer than three months, at intervals of three months after the first day thereof.
  - 2. Requirements of Law.
  - (a) In the event that any Regulatory Change shall:
- (i) change the basis of taxation of any amounts payable to any Lender under this Agreement or either Note in respect of any Libor Loans (other than taxes imposed on the overall net income of the Bank);
- (ii) impose or modify any reserve, compulsory loan assessment, special deposit or similar requirement relating to any extensions of credit or other assets of, or any deposits with or other liabilities of, any office of the Bank (including any of such Libor Loans or any deposits referred to in the definition of "Libor Base Rate"); or
- (iii) impose any other conditions affecting this Agreement in respect of Libor Loans (or any of such extensions of credit, assets, deposits or liabilities);

and the result of any of the foregoing shall be to increase the Bank's costs of making or maintaining any Libor Loans, or any Commitment, or to reduce any amount receivable by the Bank hereunder in respect of any of its Libor Loans or any Commitment, in each case only to the extent that such additional amounts are not included in the Libor Base Rate applicable thereto, then the Borrower shall pay on demand to the Bank from time to time, such additional amounts as shall be sufficient to compensate the Bank for such increased costs or reduced amount receivable.

- 3. Limitations on Libor Loans; Illegality.
- (a) Anything herein to the contrary notwithstanding, if, on or prior to the determination of an interest rate for any Libor Loans for any applicable Libor Period, the Lender shall determine (which determination shall be conclusive absent manifest error) that:
  - (i) by reason of any event affecting United States money markets or the London interbank market, quotations of interest rates for the relevant deposits are not being provided in the relevant amounts or for the relevant maturities for purposes of determining the rate of interest for such Loans under this Agreement; or
  - (ii) the rates of interest referred to in the definition of "Libor Base Rate", on the basis of which the rate of interest on any Libor Loans for such period is determined, do not accurately reflect the cost to the Bank of making or maintaining such Libor Loans for such period;

then the Bank shall give the Borrowers prompt notice thereof (and shall thereafter give the Borrowers prompt notice of the cessation, if any, of such condition), and so long as such condition remains in effect, the Bank shall be under no obligation to make Libor Loans or to convert Prime Rate Loans

into Libor Loans and the Borrowers shall, on the last day(s) of the then current Libor Period(s) for any outstanding Libor Loans, either prepay such Libor Loans in accordance with Section 4 of this Schedule B or convert such Loans into Prime Rate Loans in accordance with Section 1.4.

- (b) Notwithstanding any other provision herein, if any Regulatory Change shall make it unlawful for the Bank to make or maintain Libor Loans as contemplated by this Agreement, (i) the Bank's commitment hereunder to make Libor Loans, continue Libor Loans as such and convert Prime Rate Loans to Libor Loans shall thereupon terminate and (ii) the Loans then outstanding as Libor Loans, if any, shall be converted automatically to Prime Rate Loans on the respective last days of the then current Libor Periods with respect to such Loans or within such earlier period as required by law. If any such conversion of a Libor Loan occurs on a day which is not the last day of the then current Libor Period with respect thereto, the Borrower shall pay to such Lender such amounts, if any, as may be required pursuant to Section 4 of this Schedule B.
  - 4. Payments before end of Libor Period, Etc.

The Borrowers shall pay to the Bank upon the Bank's request, such amount or amounts as shall compensate the Bank for any loss (including loss of profit), cost or expense incurred by the Bank (as reasonably determined by it) as a result of:

- (a) any payment or prepayment or conversion of any Libor Loan on a date other than the last day of the Libor Period for such Libor Loan (including without limitation any such payment, prepayment or conversion required under Section 3 of this Schedule B); or
- (b) any failure by the Borrowers to borrow, convert into or continue a Libor Loan on the date for such borrowing specified in the relevant Notice of Borrowing or Conversion or otherwise; such compensation to include, without limitation, an amount equal to: (i) any loss or expense suffered by the Bank during the period from the date of receipt of such early payment or prepayment or the date of such conversion or failure to borrow, convert or continue to the last day of such Libor Period, if the rate of interest obtainable by the Bank upon the redeployment of an amount of funds equal to the Libor Loans so paid, prepaid or converted or as to which such failure to borrow, convert or continue applies is less than the rate of interest applicable to such Libor Loans for such Libor Period, and (ii) any loss or expense suffered by the Bank in liquidating Libor deposits prior to maturity which the Bank is unable to redeploy and which correspond to the Libor Loans so paid, prepaid or converted or as to which such failure to borrow, convert or continue applies. The determination by the Bank of the amount of any such loss or expense, when set forth in a written notice to the Borrowers, containing the Bank's good faith calculation thereof in reasonable detail, shall be presumed correct in the absence of manifest error.

EXHIBIT 1.2(a)

REVOLVING CREDIT NOTE

\$15,000,000 Massachusetts Boston,

March 21, 1997

Massachusetts corporation, and BOSTON BEER COMPANY LIMITED PARTNERSHIP, a Massachusetts limited partnership (the "Borrowers"), hereby jointly and severally promise to pay to FLEET NATIONAL BANK (the "Bank"), or order, the principal amount of \$15,000,000, or, if less, the aggregate unpaid principal amount of Revolving Loans A hereunder made by the Bank to the Borrowers under the Credit Agreement of even date herewith by and among the Borrowers and the Bank (as amended or extended from time to time, the "Loan Agreement"), together with interest thereon at the rate provided in the Loan Agreement, payable on the dates and in such amounts as specified in the Loan Agreement, and at the final maturity of this Note, whether by payment or prepayment, acceleration or otherwise, without set-off, deduction or counterclaim. The principal amount hereof may be advanced, repaid and readvanced in accordance with the terms of the Loan Agreement.

Overdue principal (whether on the due date of a Loan, by reason of acceleration or otherwise) and, to the extent permitted by applicable law, overdue interest and fees or any other amounts payable under the Loan Agreement (including without limitation overadvances) due to the Borrowers' failure to pay the same in full shall bear interest from and including the due date thereof until paid, at a rate per annum equal to 3% above the rate then applicable to Prime Rate Loans, which interest shall be compounded daily and payable on demand. In addition, if the entire amount of any required principal and/or interest is not paid in full within ten days after the same is due, the Borrowers shall pay to the Bank a late fee equal to 5% of the required payment.

The outstanding principal amount of this Note and accrued and unpaid interest thereon shall be due and payable as provided in the Loan Agreement. The Borrowers have the right in certain circumstances to prepay this Note in whole or in part subject to the terms and conditions provided in the Loan Agreement.

All payments under this Note shall be made at the head office of the Bank at One Federal Street, Boston, Massachusetts 02110 (or at such other place as the Bank may designate to Borrower from time to time in writing) in lawful money of the United States of America in federal or other immediately available funds.

This Note is the "Revolving Credit Note" referred to in and is entitled to the benefits of the Loan Agreement (including the Schedules thereto) and all other instruments evidencing and/or securing the indebtedness hereunder, which Loan Agreement and other instruments are hereby made part of this Note and are deemed incorporated herein in full; but neither this reference to the Loan Agreement nor any provision thereof shall affect or impair the absolute and unconditional obligation of the Borrower to pay the principal of and interest on this Note as herein provided. The occurrence or existence of an "Event of Default" as defined in the Loan Agreement shall constitute a default under this Note and shall entitle the Bank to accelerate the entire indebtedness hereunder and to take such other action as may be provided for in the Loan Agreement or any other instrument or agreement evidencing and/or securing this Note.

The Borrowers hereby waives presentment, demand, notice of dishonor, protest and all other demands and notices in connection with the delivery, acceptance, performance and enforcement of this Note.

All agreements among the Borrowers and the Bank are hereby expressly limited so that in no contingency or event whatsoever, whether by reason of acceleration of maturity of the indebtedness

or otherwise, shall the amount paid or agreed to be paid for the use or forbearance of the indebtedness evidenced hereby exceed the maximum amount which the Bank is permitted to receive under applicable law. If, from any circumstances whatsoever, fulfillment of any provision hereof or of the Loan Agreement, at the time performance of such provision shall be due, shall involve exceeding such amount, then the obligation to be fulfilled shall automatically be reduced to the limit of such validity and if, from any circumstances the Bank should ever receive as interest an amount which would exceed such maximum amount, such amount which would be excessive interest shall be applied to the reduction of the principal balance evidenced hereby and not to the payment of interest. As used herein, the term "applicable law" shall mean the law in effect as of the date hereof, provided, however, that in the event there is a change in the law which results in a higher permissible rate of interest, then this Note shall be governed by such new law as of its effective date. This provision shall control every other provision of all agreements among the Borrowers and the Bank.

If this Note shall not be paid when due and shall be placed by the holder hereof in the hands of any attorney for collection, through legal proceedings or otherwise, the Borrowers will pay reasonable attorneys' fees to the holder hereof together with reasonable costs and expenses of collection, including, without limitation, any such reasonable attorneys' fees, costs and expenses relating to any proceedings with respect to the bankruptcy, reorganization, insolvency, readjustment of debt, dissolution or liquidation of either of the Borrowers or any party to any instrument or agreement securing this Note, all as provided in the Loan Agreement.

This instrument shall have the effect of an instrument executed under seal and shall be governed by and construed in accordance with the laws of the Commonwealth of Massachusetts (without giving effect to any conflicts of laws provisions contained therein).

THE BOSTON BEER COMPANY, INC.

By: Title:

BOSTON BEER COMPANY LIMITED PARTNERSHIP

By The Boston Brewing Company, Inc.,
Its General Partner

By: Title:

EXHIBIT 1.2(b)

REVOLVING CREDIT AND TERM NOTE

\$30,000,000

Boston, Massachusetts

March 21, 1997

Massachusetts corporation, and BOSTON BEER COMPANY LIMITED PARTNERSHIP, a Massachusetts limited partnership (collectively, the "Borrowers"), hereby jointly and severally promise to pay to FLEET NATIONAL BANK (the "Bank"), or order, the principal amount of \$30,000,000, or, if less, the aggregate unpaid principal amount of Revolving Loans B hereunder made by the Bank to the Borrowers under the Credit Agreement of even date herewith by and among the Borrowers and the Bank (as amended or extended from time to time, the "Loan Agreement"), together with interest thereon at the rate provided in the Loan Agreement, payable on the dates and in such amounts as specified in the Loan Agreement, and at the final maturity of this Note, whether by payment or prepayment, acceleration or otherwise, without set-off, deduction or counterclaim. The principal amount hereof may be advanced, repaid and readvanced in accordance with the terms of the Loan Agreement.

Overdue principal (whether on the due date of a Loan, by reason of acceleration or otherwise) and, to the extent permitted by applicable law, overdue interest and fees or any other amounts payable under the Loan Agreement (including without limitation overadvances) due to the Borrowers' failure to pay the same in full shall bear interest from and including the due date thereof until paid, at a rate per annum equal to 3% above the rate then applicable to Prime Rate Loans, which interest shall be compounded daily and payable on demand. In addition, if the entire amount of any required principal and/or interest is not paid in full within ten days after the same is due, the Borrowers shall pay to the Bank a late fee equal to 5% of the required payment.

The outstanding principal amount of this Note and accrued and unpaid interest thereon shall be due and payable as provided in the Loan Agreement. The Borrowers have the right in certain circumstances to prepay this Note in whole or in part subject to the terms and conditions provided in the Loan Agreement.

All payments under this Note shall be made at the head office of the Bank at One Federal Street, Boston, Massachusetts 02110 (or at such other place as the Bank may designate to Borrower from time to time in writing) in lawful money of the United States of America in federal or other immediately available funds.

This Note is the "Revolving Credit and Term Note" referred to in and is entitled to the benefits of the Loan Agreement (including the Schedules thereto) and all other instruments evidencing and/or securing the indebtedness hereunder, which Loan Agreement and other instruments are hereby made part of this Note and are deemed incorporated herein in full; but neither this reference to the Loan Agreement nor any provision thereof shall affect or impair the absolute and unconditional obligation of the Borrower to pay the principal of and interest on this Note as herein provided. The occurrence or existence of an "Event of Default" as defined in the Loan Agreement shall constitute a default under this Note and shall entitle the Bank to accelerate the entire indebtedness hereunder and to take such other action as may be provided for in the Loan Agreement or any other instrument or agreement evidencing and/or securing this Note.

The Borrowers hereby waives presentment, demand, notice of dishonor, protest and all other demands and notices in connection with the delivery, acceptance, performance and enforcement of this Note.

All agreements among the Borrowers and the Bank are hereby expressly limited so that in no contingency or event whatsoever, whether by reason of acceleration of maturity of the indebtedness

or otherwise, shall the amount paid or agreed to be paid for the use or forbearance of the indebtedness evidenced hereby exceed the maximum amount which the Bank is permitted to receive under applicable law. If, from any circumstances whatsoever, fulfillment of any provision hereof or of the Loan Agreement, at the time performance of such provision shall be due, shall involve exceeding such amount, then the obligation to be fulfilled shall automatically be reduced to the limit of such validity and if, from any circumstances the Bank should ever receive as interest an amount which would exceed such maximum amount, such amount which would be excessive interest shall be applied to the reduction of the principal balance evidenced hereby and not to the payment of interest. As used herein, the term "applicable law" shall mean the law in effect as of the date hereof, provided, however, that in the event there is a change in the law which results in a higher permissible rate of interest, then this Note shall be governed by such new law as of its effective date. This provision shall control every other provision of all agreements among the Borrowers and the Bank.

If this Note shall not be paid when due and shall be placed by the holder hereof in the hands of any attorney for collection, through legal proceedings or otherwise, the Borrowers will pay reasonable attorneys' fees to the holder hereof together with reasonable costs and expenses of collection, including, without limitation, any such reasonable attorneys' fees, costs and expenses relating to any proceedings with respect to the bankruptcy, reorganization, insolvency, readjustment of debt, dissolution or liquidation of either of the Borrowers or any party to any instrument or agreement securing this Note, all as provided in the Loan Agreement.

This instrument shall have the effect of an instrument executed under seal and shall be governed by and construed in accordance with the laws of the Commonwealth of Massachusetts (without giving effect to any conflicts of laws provisions contained therein).

THE BOSTON BEER COMPANY, INC.

By: Title:

BOSTON BEER COMPANY LIMITED PARTNERSHIP

By The Boston Brewing Company,
Inc.,
 Its General Partner

By: Title:

EXHIBIT 1.4

NOTICE OF BORROWING OR CONVERSION

Fleet National Bank One Federal Street Boston, MA 02110

Date:

Attn: Mary Barcus, Vice President

Re: Credit Agreement dated as of March 21, 1997 (as amended from time to time, the "Agreement") among Fleet National Bank and The Boston Beer Company, Inc. and Boston Beer Company Limited Partnership (collectively, the "Borrowers")

Dear Ms. Barcus:

continuation.1

Please accept this notice as the election of the Borrowers as follows (Libor Loan must be in increments of \$100,000 and equal to or greater than \$100,000):

A new [Revolving Loan A/Revolver Loan B] request in the amount of \$ to be borrowed as a [Prime Rate Loan/Libor Loan]. [Such Libor Rate option is selected for a term of:

one month
two months
three months from the date of the Loan advance.]

Conversion or continuation of a portion of outstanding [Revolving Loans A/Revolving Loans B/Term Loan] in the amount of \$ from a [Prime Rate Loan/Libor Loan] to a [Prime Rate Loan/Libor Loan]. [Such Libor Rate option is selected for a term of:

one month two months three months from the date of the conversion or

This Notice is issued in accordance with the provisions of Section 1.4 of the Loan Agreement and is subject to the terms of the Loan Agreement.

The representations and warranties contained in the Agreement are true and accurate on and as of the effective date of such advance, conversion or continuation as though made at and as of such date (except to the extent that such representations and warranties expressly relate to an earlier date); and no Default has occurred and is continuing or will result from such advance, conversion or continuance. The undersigned represents and warrants that he is authorized to execute and deliver this Notice on behalf of the Borrowers.

THE BOSTON BEER COMPANY, INC. and BOSTON BEER COMPANY LIMITED PARTNERSHIP

By:

Duly authorized signatory as to both

EXHIBIT 2.1

\$45,000,000 Revolving Credit and Term Financing

of

THE BOSTON BEER COMPANY, INC.

and

## FLEET NATIONAL BANK

March 21, 1997

## AGENDA OF CLOSING

		Responsible Counsel
1.	Credit Agreement between the Bank and the Borrowers	Agreement: E&A Disclosure Exhibits: HWD
2.	\$30,000,000 Revolving Credit and Term Note	E&A
3.	\$15,000,000 Revolving Credit Note	E&A
4.	Guaranty of Samuel Adams Brewing Company, Ltd.	E&A
5.	Certificate of Insurance	HWD
6.	Officer's Certificate certifying as to (and attaching) the Partnership's (a) certificate of limited partnership, (b) partnership agreement and (c) partners' votes authorizing all transactions contemplated by the Loan Documents	HWD
7.	Officer's Certificates certifying as to (and attaching) the Corporation's, the Material Subsidiary's and the General Partner's (a) corporate charter, (b) bylaws, and (c) director's votes authorizing all transactions contemplated by the Loan Documents; and certifying as to the incumbency of the officers signing on behalf of the Borrowers and the Material Subsidiary	HWD
8.	Certificates of legal existence and partnership or corporate good standing of the Borrowers, the General Partner and the Material Subsidiary	HWD
9.	UCC and federal tax lien searches against the Borrowers and the Material Subsidiary (including trade names) and their predecessors (within 5 years)	HWD
10.	Copy of IRB Agreement and other Material Agreements, if any	HWD
11.	Opinion of General Counsel to the Borrowers and the Material Subsidiary	HWD

## ABBREVIATIONS

Borrowers	The Boston Beer Company, Inc. ("Corporation") and Boston Beer Company Limited Partnership ("Partnership")
Bank	Fleet National Bank
E&A	Edwards & Angell, special counsel to
	the Bank
HWD	Hutchins, Wheeler & Dittmar
General Partner	The general partner of the Partnership
Material	Samuel Adams Brewery Company, Ltd.
Subsidiary	

## Litigation

Premier Worldwide Beer ("Premier"), the Partnership's former exclusive distributor in the United Kingdom (the "U.K.") has brought suit in the U.K. against the Partnership and Whitebread PLC, asserting that certain aspects of the relationship between Whitebread PLC and the Partnership which commenced prior to the expiration of Premier's distribution rights violated those rights. The Borrowers currently believe that the matter can be settled with Borrowers' share of any settlement payment to Premier not exceeding U.S. \$3,000,000.

#### EXHIBIT 4.1

Officer's Compliance Certificate for Period Ended []

BOSTON BEER COMPANY LIMITED PARTNERSHIP and THE BOSTON BEER COMPANY, INC. (collectively, the "Borrowers") HEREBY CERTIFIES that:

This Report is furnished pursuant to Section 4.1(c) of the Credit Agreement dated as of March 21, 1997 by and among the Borrowers and Fleet National Bank, as amended from time to time (the "Agreement"). Unless otherwise defined herein, the terms used in this Report have the meanings assigned to them in the Agreement.

As required by Section 4.1[(a)] or [(b)] of the Agreement, the Borrowers' financial statements for the [year/quarter] ended , 199 (the "Financial Statements"), prepared in accordance with generally accepted accounting principles consistently applied, accompany this Report. The Financial Statements present fairly the financial position of the Borrowers and their Subsidiaries as at the date thereof and the results of operations for the period covered thereby [(subject only to normal recurring year-end adjustments)].

Based on the Financial Statements provided with this Report, the figures set forth in Schedule A hereto for determining compliance by the Borrowers with the financial covenants contained in Section 5 of the Agreement for the applicable reporting periods are true, complete and correct.

The activities of the Borrowers during the periods covered by such Financial Statements have been reviewed by the Chief Financial Officer of the Borrowers or by employees or agents under his or her immediate supervision. Based on such review, to the best knowledge and belief of the undersigned, and as of the date of this Report, the representations and warranties contained in the Agreement are true and accurate as of the date hereof (except to the extent that such representations and warranties relate to an earlier date) and no Default has occurred.\*

WITNESS my hand this

day of

, 19\_\_.

BOSTON BEER COMPANY LIMITED PARTNERSHIP and THE BOSTON BEER COMPANY, INC.

By:

\* If a Default has occurred as of or prior to such date, this paragraph is to be modified with an appropriate statement as to the nature thereof, the period of existence thereof and what action the Borrowers have taken, is taking, or proposes to take with respect thereto.

# Schedule A to Compliance Certificate for Period Ended []

## 1. MINIMUM TANGIBLE NET WORTH (Section 5.1(a))

1-	١.	7 0 + 11 0 7	manaible	NT o +	TaT o ao + b	o f	+ h o	Commomntion
( a	. )	ACLUAL	Tangible	Net	WOLLII	OT	LHE	Corporation

(i <sub>)</sub>	Total	Assets		\$
(ii)	minus	goodwill		\$
(iii)	minus	intangible	items	\$

(lV)	mınus	reserves	not	
	alread	ly deducte	ed	:

(V)	minus	write	-up	o in	
	book v	alue	of	assets	\$

- (vi) minus value of minority interests
   (excluding Permitted
   Acquisitions)
- (vii) minus Total Liabilities \$
- Actual Tangible Net Worth \$
- (b) Minimum Tangible Net Worth required (\$50,000,000 plus 50% positive Net Income step up in each fiscal quarter)

## 2. LEVERAGE (Section 5.1(b))

- (a) Actual Ratio of Total Liabilities to Tangible Net Worth:
  - (i) Total Liabilities \$
  - (ii) Tangible Net Worth
  - (iii) Actual Ratio of Total
     Liabilities to Tangible Net
     Worth :1.00
- (b) Maximum Ratio of Total Liabilities to Tangible Net Worth permitted 1.00:1.00
- 3. TOTAL FUNDED DEBT/EBITDA RATIO
   (Section 5.1(c)) (rolling four-quarter
   test)
  - (a) Actual Ratio of Total Funded Debt to EBITDA:
    - (i) Actual Total Funded Debt (as

			of la	ast day of period)	\$
		(ii)	EBITI	DA:	
			(1)	Net Income	\$
			(2)	plus taxes in respect of income and profits	\$
			(3)	plus Interest Expense	\$
			(4)	plus depreciation, amoritization and non-	
				cash charges	\$
			EBITI	DA Total	\$
		(iii)		al Ratio of Total Funded to EBITDA	2.50:1.00
	(b)	Maxim to EB	um Rat ITDA p		
4.	PROF	ITABIL	ITY (S	Section 5.1(d))	
	(a)	Actua	l prof	its for fiscal quarter	\$
	(b)		um req 1 quar	\$1.00	